### Part II

## China's Role in the Region and in the Global Financial System

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, <u>www.fondad.org</u>

## 6

### **Chiang Mai and Beyond**

Yung Chul Park and Yunjong Wang<sup>1</sup>

Defore the Asian financial crisis broke out in 1997, few would Dhave seriously argued for the creation of a system of regional financial cooperation in Asia. Only market-led natural economic territories were being formed in East Asia, often cutting across political lines (Scalapino, 1991). However, the financial crisis that erupted in 1997 was a major financial breakdown that gave Asians a strong impetus to search for a regional arrangement that could forestall future crises. For instance, Japan proposed the creation of an Asian Monetary Fund (AMF) as a framework for financial cooperation and policy coordination in the region. A regional monetary fund, it was argued, would provide a means of defense, in addition to the IMF lending facilities, against financial crises in Asia. Although the proposal received a positive response from a number of Asian countries, it was shot down by the objection of the US, EU, and the IMF as well as the lukewarm attitude of a few Asian countries in the region.<sup>2</sup>

The idea of an AMF was revived when the finance ministers of China, Japan, and South Korea, along with the ten members of the Association of South-East Asian Nations (ASEAN) agreed on May 6, 2000 in Chiang Mai, Thailand to establish a system of swap

<sup>&</sup>lt;sup>1</sup> Comments by Charles Adams and participants to the conference are gratefully acknowledged.

<sup>&</sup>lt;sup>2</sup> See, for an insider's account of the AMF's killing, Eisuke Sakakibara's remarks in: Jan Joost Teunissen and Mark Teunissen (eds.), *Financial Stability and Growth in Emerging Economies: The Role of the Financial Sector*, Fondad, The Hague, 2003, p. 240.

arrangements within the group.<sup>3</sup> This regional scheme for financial cooperation – known as the Chiang Mai Initiative (CMI) – is now gathering momentum and opening the doors to possibly significant policy-led integration in East Asia.

Since then, the deputy financial ministers of the thirteen ASEAN+3 countries have negotiated the details of the initiative to produce a basic framework of the ASEAN Swap Arrangement (ASA) and Bilateral Swap Arrangements (BSAs) and repurchase agreement. The framework was approved at the meeting of the deputies on November 7, 2000 in Beijing. A progress report on the CMI was then presented at the summit meeting of the thirteen countries two weeks later.

The CMI swap arrangements have been designed and implemented to provide liquidity support to member countries that experience short-run balance of payment deficits. By preventing an extreme crisis or systemic failure in a country and subsequent regional contagion as occurred in the recent Asian financial crisis, this facility is expected to contribute to financial stability in the region. However, emergency support facilities such as the CMI, which are similar in nature to other regional and international "lender of last resort" facilities, are primarily for systemic purposes and as such would likely be used very infrequently. Since the intent of the CMI is to be proactive, there is a need to define a mutually agreed framework for inter-governmental cooperation amongst the ASEAN+3, that can quickly and effectively implement emergency assistance at required levels when the need arises. Moreover, a group approach would ensure that any conditionality associated with the financial assistance is consistently applied across countries.

Although the adoption and implementation of the CMI could be counted as a major step toward strengthening financial cooperation among the thirteen ASEAN+3 countries, these countries will face much tougher challenges and tasks in exploring developments beyond the CMI. East Asian countries need to clarify to the international community what their motivations are, how they will develop an action plan, and how they believe it fits in with the

<sup>&</sup>lt;sup>3</sup> The ASEAN members are Brunei, Darussalam, Cambodia, Indonesia, Laos, Malaysia, Myanmar, the Philippines, Singapore, Thailand, and Vietnam. The group of China, Japan, and South Korea, along with the ten members of the ASEAN is known as "ASEAN+3".

existing global financial system. And furthermore, it remains to be seen whether East Asia can emulate the policy-led economic integration that took place in Europe through this kind of financial cooperation framework.

At present, the CMI does not require a new institution like the proposed AMF, and it is also tightly linked to IMF conditionalities. At this stage of development, East Asians may not be well prepared to negotiate an international agreement that includes provisions for sanctions and fines for countries that do not adjust their domestic policies accordingly.<sup>4</sup> In this regard, the CMI and its follow-up implementation are acceptable to many detractors of a regional monetary fund. At the most elementary stage of zero institutional integration, the moral hazard problem associated with mutual liquidity provisions can be addressed by the linkage of the CMI with IMF conditionalities (Wang and Woo, 2002). As long as the CMI is simply a source of financial resources supplementary to the IMF, the size of the swap borrowing does not have to be sufficient to meet potential needs, because members can always rely on the deep pocket of financial resources provided by the IMF (Wang, 2001).

Under the CMI framework, East Asian countries have not yet specified common policy objectives toward economic integration. Joint efforts for crisis prevention in the region are rather ambiguous as a policy objective toward promoting economic integration in the region. In this regard, the CMI lies between cooperation and coordination, but is closer to cooperation on the spectrum of economic integration. Two main components are being implemented under the ASEAN+3 framework. One is a regional financial facility under the CMI, while the other is the development of a regional surveillance and policy dialogue mechanism to complement the CMI. However, neither of these components has yet been com-

<sup>&</sup>lt;sup>4</sup> For instance, the ASEAN surveillance process is built on the basis of consensus and informality in keeping with the tradition of non-interference (Manzano, 2001). East Asia, in contrast to Europe, lacks the tradition of integrationist thinking and the web of interlocking agreements that encourage monetary and financial cooperation (Eichengreen and Bayoumi, 1999). Eichengreen and Bayoumi (1999) stress that East Asia does not meet the necessary intellectual preconditions for regional integration. For this reason, they conclude that it is unrealistic to speak of pooling national sovereignties. While there is no doubt considerable work to be done in promoting policy coordination in the region, it is wrong to say that it cannot be done in East Asia.

pleted and many members feel they are still in need of significant amendments. The BSA under the CMI will be reviewed in 2004. At that time, the ASEAN+3 may decide to maintain, amend or abolish the current arrangements. The various modalities of a regional surveillance mechanism are now under discussion, but it is highly unlikely that an effective surveillance mechanism will be put in place any time soon.

The purpose of this paper is to provide a view on the current process and future prospects for regional financial and monetary cooperation in East Asia. Looking forward, financial cooperation in East Asia will be evolutionary. For over a half century, European countries worked hard to develop various institutional frameworks which encouraged their cooperation on monetary and financial matters. If the European experience is any guide, it may take many years to develop effective cooperative arrangements and institutions in East Asia. However, some observers note that East Asia may be on the brink of a historical evolution, as Europe was half a century ago (Bergsten, 2000; Dieter, 2000). When the regional group agrees on deepening regional integration through financial and monetary cooperation, it remains to be seen whether East Asia can successfully emulate the European integration process through solid institution building.

The remainder of this paper is organised as follows. Section 1 provides an overview of the current structure of the CMI and discusses a number of its pitfalls. Section 2 explores the issues related to the creation of a monitoring and surveillance mechanism. Section 3 addresses major barriers to financial cooperation and integration. Finally, Section 4 concludes with a discussion of the future prospects for financial and monetary cooperation in the region.

#### 1 Overview of the Chiang Mai Initiative (CMI)

The Chiang Mai Initiative (CMI) has two components: (i) an expanded ASEAN Swap Arrangement (hereafter ASA) encompassing the ten ASEAN countries; and (ii) a network of Bilateral Swap Arrangements (hereafter BSA) and repurchase arrangements basically encompassing the thirteen ASEAN+3 countries.

In 1977, the five original ASEAN countries – Indonesia, Malaysia, Philippines, Singapore, and Thailand – agreed to establish

the ASA in pursuit of their common objective of promoting financial cooperation. Since then, the ASA has been renewed several times, but it has remained a very primitive financial arrangement, mainly due to the loose financial cooperation in ASEAN. The utilisation was very low. Even during the Asian financial crisis, countries did not turn to the ASA.

An expanded ASA became effective in November 2000 to include the five new members of ASEAN, and the total amount of the facility was raised to \$1 billion from the initial amount of \$200 million. The currencies available under the ASA are the US dollar, yen, and euro. The euro, yen and Euro LIBOR interest rates are used as the base rates for swap transactions. Each member is allowed to draw from the facility a maximum of twice its committed amount for a period not exceeding six months, subject to an extension for another period not exceeding six months.

A major drawback of the ASA stems from the "equal partnership" condition, which stipulates that the amount to be granted to a swaprequesting member country shall be provided by the other member countries in equal shares. In addition, a participant may refrain from providing the committed lending by merely informing the other member countries of its decision, and may, at its discretion, provide reasons for the decision. Under such circumstances, other participants, on a voluntary basis, can increase their shares. If the total amount of swap committed collectively by the participants does not sufficiently meet the requested amount, the amount of swap granted is to be reduced accordingly. Considering these conditions, the ASA would not help much to minimise the disruption caused by a financial meltdown in the event that massive scale liquidity provisions are required.

If ASEAN members fail to find a country to contribute a meaningful amount of credit, they will have to link the ASA to global liquidity facilities provided by the IMF or other regional facilities. The ASA could be further expanded to encompass the three North-East Asian countries – China, Japan, and South Korea. However, as long as the two problems of disproportionate contribution obligations and non-compulsory commitments remain unsolved, the effectiveness of the ASA may be seriously undermined. Major revisions should be undertaken to secure the firm commitment of participating countries and enlarged contributions. Alternatively, the second component of the CMI – the network of

BSA – could be a useful source of complementary financial resources.

The BSA is a facility that provides short-term liquidity assistance in the form of swaps of US dollars with the domestic currencies of participating countries.<sup>5</sup> The maximum amount that can be drawn under each BSA is to be determined through bilateral negotiations. However, disbursements to a member in need of liquidity assistance are expected to be made in a concerted manner through consultation among the swap-providing countries with one of the swap-providing countries serving as the coordinator for the consulting process. The BSA allows a disbursement of up to ten percent of the maximum amount of drawing without an agreement, or even a prospective agreement, with the IMF. Such a disbursement would represent "short-term liquidity support", and would be renewable only once, and thus repayable within six months of the original transaction. However, countries drawing more than the ten percent are required to accept an IMF programme of macroeconomic and structural adjustments. In this sense, the BSA is complementary to the IMF's financial assistance.

A number of the participating countries have expressed reservations about the linkage of the BSA with IMF conditionalities and have proposed to increase gradually the ten percent limit on independent disbursements and also abolish the IMF linkage after a period of transition. For instance, Malaysia advocates complete independence of the CMI from the IMF. Severance of the IMF linkage requires the creation of a regional surveillance mechanism for the CMI. At the fifth ASEAN finance ministers' meeting in April 2001 in Kuala Lumpur, however, there was consensus that the BSA should remain complementary and supplementary to the IMF facilities until a regional surveillance system is established. The ASEAN ministers also agreed that "the terms and modalities of the BSA should take into account the different economic fundamentals, specific circumstances, and financing needs of individual countries". This agreement implies that the contracting parties of the BSA could deviate from the basic framework on setting terms and conditions of the swap agreements.

Participating countries are able to draw from the BSA for a

 $<sup>^{\</sup>scriptscriptstyle 5}$  The one exception is the Japan-China BSA, which would swap yen for renminbi.

period of 90 days. The first drawing may be renewed seven times. The interest rate applicable to the drawing is the LIBOR plus a premium of 150 basis points for the first and first renewal drawings. Thereafter, the premium is increased by an additional 50 basis points for every two renewals, but not exceeding 300 basis points.

A repurchase agreement was also established to provide shortterm liquidity to a participating member through the sale and buyback of appropriate securities. Securities of the repurchase agreement are US Treasury notes or bills with a remaining maturity of not more than five years and government securities of the counterparty country. The period of the repurchase agreement is one week, but could be extended on the termination value date by agreement between the contracting parties. The minimum amount for each transaction requested is five percent of the total amount of the repurchase agreement. In each transaction, the buyer is given a margin of 102 percent for US Treasury notes or bills and 105 percent for government securities of the counterparty country. At the inception of the CMI, the participating countries considered expanding these facilities through bilateral negotiations between the contracting parties. However, no new repurchase agreements have been made since the Chiang Mai meeting. Instead, ASEAN+3 governments have focused on negotiating and concluding BSAs.

Since the ASEAN+3 summit meeting in November 2000, Japan, China, and South Korea have been negotiating BSAs with each other and with the ASEAN countries. Japan has been most active: it has concluded six agreements with China, South Korea, Indonesia, Malaysia, the Philippines and Thailand, and is currently negotiating an agreement with Singapore. Korea also concluded four agreements with China, Malaysia, the Philippines and Thailand in addition to the Japan-Korea BSA. Similarly, China also concluded three agreements with Malaysia, the Philippines and Thailand in addition to its agreements with Japan and Korea.

From the beginning, Singapore and Brunei have shown little enthusiasm for the CMI, largely because they believe the BSAs with their neighbouring countries will be one-way arrangements in which they will be asked to provide a large amount of liquidity in case of a crisis affecting the ASEAN region. However, Japan has made progress in bringing Singapore to the negotiating table by proposing a BSA that uses local currencies rather than the US dollar. In fact, Japan has concluded a similar local-currency BSA with China. Indonesia has not shown any strong interest in negotiating BSA arrangements because of its preoccupation with domestic economic issues and the management of its huge foreign debts, not to mention escalating political instability. Recently, Indonesia has concluded a BSA with Japan, though it does not appear to place a high priority on contracting additional BSAs with other members of the CMI.

At present, the total amount of BSAs covering all thirteen countries is estimated to be around \$32 billion (see Table 1). The maximum amount any individual country can draw varies a great deal. In the case of Thailand, the maximum is about \$6 billion. Given such a relatively small amount of liquidity available through the CMI, doubts have been raised as to whether the BSA system could truly be a credible and effective system of defense against future speculative attacks. Participants of international financial markets are not likely to be impressed with the amount of liquidity available and hence ignore the CMI, unless the ASEAN+3 are prepared to increase the number of BSAs and expand the swap amount of each BSA. However, financial significance does not matter as long as the CMI maintains the linkage with the IMF. If the CMI goes beyond its supplementary role to the IMF by establishing a regional fund and seeking independent conditionalities, then its financial position will have to be significantly enlarged so it can effectively prevent financial crises in the region.

In comparison with the European facilities, the CMI has had a different motivation from the outset. The European facilities were created with the purpose of limiting exchange rate fluctuations under a coordinated exchange rate mechanism. The CMI started with high capital mobility and flexible exchange rates, although some members of ASEAN+3 have maintained relatively fixed exchange rate regimes. Even now, the CMI does not presume any manifest exchange rate coordination.

#### 2 Creation of Monitoring and Surveillance System

From the inception of the CMI, some member countries have opposed the idea of linking the CMI with the IMF programme. Other members, in particular Japan and China, have argued for the importance of forging a cooperative relation with the IMF at an early stage of CMI development to enhance its credibility. They

BSA	Currencies	Conclusion Dates	Size
I IZ	ф <b>ЛТ</b> 7	411 2001	<u> </u>
Japan-Korea	\$/Won	4 July 2001	\$/ billion*
Japan-Thailand	\$/Baht	30 July 2001	\$3 billion
Japan-Philippines	\$/Peso	27 August 2001	\$3 billion
Japan-Malaysia	\$/Ringgit	5 October 2001	\$3.5 billion*
Japan-PRC	Yen/Renmimbi	28 March 2002	\$3 billion
Japan-Indonesia	\$/Rupiah	17 February 2003	\$3 billion
Korea-PRC	Won/Renmimbi	24 June 2002	\$2 billion
Korea-Thailand	\$/local currency	25 June 2002	\$1 billion
Korea-Malaysia	\$/local currency	26 July 2002	\$1 billion
Korea-Philippines	\$/local currency	9 August 2002	\$1 billion
PRC-Thailand	\$/Baht	6 December 2001	\$2 billion
PRC-Malaysia	\$/Ringgit	9 October 2002	\$2 billion
PRC-Philippines	\$/Peso	Negotiation completed	\$2 billion
Japan-Singapore		Under negotiation	

#### Table 1 Progress on the Chiang Mai Initiative

(in billions of dollars, as of March 31, 2003)

Notes:

BSA = Bilateral Swap Arrangement; PRC = People's Republic of China; and, Korea = Republic of Korea

\* The US dollar amounts include the amounts committed under the New Miyazawa Initiative: \$5 billion for Korea and \$2.5 billion for Malaysia.

have succeeded in persuading Malaysia and other opposing members to accept the linkage of the BSAs with the IMF conditionality as a temporary scheme until a formal surveillance mechanism is put in place. Malaysia agreed to the IMF linkage with the condition of establishing a study group to examine the types of monitoring and surveillance systems the CMI would require to function as an independent regional financial arrangement.

Most participating countries agree in principle that the CMI needs to be supported by a surveillance system that (i) monitors economic developments in the region, (ii) serves as an institutional framework for policy dialogue and coordination among the members, and (iii) imposes structural and policy reform on the countries drawing from the BSAs. The ASEAN+3 finance ministers agreed to organise a study group to produce a blueprint for an effective mechanism of policy dialogues and economic reviews for the CMI operations at the ADB annual meeting in Honolulu on May 9, 2001. Japan and Malaysia were chosen to co-chair the group. The study group met in Kuala Lumpur on November 22, 2001 to

discuss the report on possible modalities of surveillance prepared by Bank Negara Malaysia and Japan's Ministry of Finance. However, the member countries could not reach an agreement on the surveillance issues, agreeing only to institutionalising the ASEAN+3 meetings of deputies for informal policy reviews and dialogues. At this stage of development of the CMI, many countries feel uncomfortable about creating an independent regional monitoring and surveillance unit as part of the CMI.

In the long run, however, the participating countries are likely to wean themselves from their reliance on the IMF. If the CMI develops into an independent regional financial arrangement from the IMF, the architect of the CMI will have to decide whether the arrangement could be supported by a surveillance mechanism based on peer reviews and pressure instead of formal policy conditionalities. In our view, economic policy dialogues and peer monitoring may not provide an institutional framework that can minimise the moral hazard problem. In this regard, it is worthwhile to distinguish conceptually two different types of moral hazards in conjunction with regional financial arrangements and related surveillance processes. One is related to liquidity assistance, while the other is related to collective actions required for common policy objectives. Peer pressure may not be effective in rectifying the moral hazard related to liquidity assistance and may have to be supplemented by surveillance and conditionalities attached to the liquidity provision. If the CMI develops into more or less an independent financial arrangement from the IMF, then the regional financial arrangement should be designed to discipline the borrowers to adhere to sound macroeconomic and financial policies by imposing conditionalities.

On the other hand, peer pressure may be an effective means of achieving the objective of economic policy coordination among members. The surveillance and policy coordination may have a double-decker structure in policy formation and implementation. Under the European Monetary Union (EMU), only specific common policies such as monetary and exchange rate policies are binding at the Community level, while economic policies such as budgetary and structural policies largely remain under the national sovereignty of member countries. In the European Union, the framework of broad economic policy guidelines provides a basis for policy coordination. This structured surveillance process has contributed to assessing the consistency of each member country's economic policies. The ASEAN+3 countries at the current stage do not seem well prepared for establishing a policy coordination mechanism in the surveillance process. In the case of European integration, a more effective and structured surveillance process started only when the European countries sought monetary integration in the 1990s (Wang and Yoon, 2002). The long history of European surveillance shows that an effective surveillance process cannot be achieved overnight but needs many years of constant interaction and mutual trust building. Thus, it will take more time for the ASEAN+3 countries to establish more comprehensive and structured surveillance systems like those under the EMU.

#### **3** Barriers to Financial Cooperation and Integration

East Asian policymakers who conceived the idea of the CMI would easily concede that the BSA system as it is currently structured has a long way to go before it can be accepted as an effective mechanism of defense against financial crises. Although two years have passed since the system was established in May 2000, the leaders of the CMI group have yet to produce an operational structure for BSAs, in particular a monitoring and surveillance mechanism. And it is highly unlikely that they will do so any time soon.

In the absence of a clear vision on the scope and modality of financial cooperation through the CMI, many financial market experts have expressed doubts about whether any country facing an incipient crisis could draw from the BSAs, and if they could, how much liquidity would be available. Technically speaking, swapproviding countries could refuse any support even within the ten percent drawing which is not linked to the IMF conditionality. Because this ten percent of swap can be disbursed only with the consent of swap-providing countries, these countries need to formulate their own assessments about the swap-requesting country. At present, the practices under the ASEAN+3 process cannot effectively identify emerging problems. In this regard, an independent surveillance unit as a standing secretariat to support the CMI operation must be immediately established. However, most participating countries seem to prefer a gradual approach by avoiding institutionalisation of the procedures. Negotiations for

additional BSAs and also the surveillance system may continue, but unless the deadlock over some of the pending issues on surveillance is broken, the future prospects of the CMI do not seem promising.

There are many economic, institutional, and political barriers to financial cooperation and integration in East Asia. A large number of empirical studies have shown that although the ASEAN+3 countries by no means constitute an ideal group for an optimum currency area, they are as qualified for a common currency as the members of the European Union were in the 1970s and 1980s.<sup>6</sup> These studies invariably point to a large increase in intra-regional trade in East Asia in recent years as a development conducive to the region's financial and monetary integration (see Table 2). Trade and investment liberalisation has been the driving force behind much of the increase in intra-regional trade. This increase, in turn, has had the effect of synchronising business cycles across East Asian countries, thereby producing economic conditions favourable for a currency union in the region.<sup>7</sup>

Against these trade and macroeconomic developments, financial deregulation and market opening have drawn East Asia away from regional financial integration. Financial liberalisation throughout East Asia has caused many countries to establish closer linkages with international financial markets, but not with other country markets in the region. There is no clear evidence that the financial markets of the East Asian countries are more regionally integrated. The financial markets of the European countries that maintained capital controls were much more integrated in the 1970s and 1980s than the markets of East Asian countries are at present.<sup>8</sup> This difference suggests that financial market liberalisation and opening may not speed up economic integration in East Asia (Park, 2002).

<sup>&</sup>lt;sup>6</sup> See Eichengreen and Bayoumi (1999), Baek and Song (2002) and Lee, Park and Shin (2002).

<sup>&</sup>lt;sup>7</sup> From the theoretical point of view, trade integration can lead to business cycle synchronisation in either direction – convergence or divergence. However, using the data of East Asian countries, various researches provide evidence that economic fluctuations are more synchronised as trade interdependence deepens in the region. See Choe (2001) and Loayza, Lopez and Ubide (2001). Based on a panel regression, Shin and Wang (2003) find that intra-industry trade is the major channel through which business cycles in East Asia become synchronised, although increased trade itself does not necessarily lead to close business cycle coherence.

	1980	1990	2000
	East Asia		
China	42.4	58.9	48.7
Hong Kong	46.7	60.4	64.1
Indonesia	62.6	56.8	54.4
Japan	23.8	28.0	38.1
Korea	32.7	34.2	42.2
Malaysia	49.2	55.1	56.1
Philippines	37.4	40.0	46.5
Singapore	49.4	50.7	57.5
Taiwan	34.9	43.1	50.9
Thailand	40.1	47.5	54.2
Average	41.9	47.5	51.3
	Europe		
Austria	69.8	75.6	71.0
Belgium	74.9	80.4	72.5
Denmark	74.4	75.5	74.8
Finland	57.4	66.6	61.7
France	57.8	67.1	66.7
Germany	63.9	67.4	59.1
Greece	48.2	70.1	56.1
Ireland	77.6	76.1	61.3
Italy	56.1	67.3	59.1
Netherlands	70.5	77.0	67.3
Norway	78.6	75.2	73.7
Portugal	58.7	78.5	78.6
Spain	43.2	68.2	67.8
Sweden	73.8	78.9	68.4
Switzerland	69.0	72.1	66.9
United Kingdom	55.8	62.8	57.0
Average	64.4	72.4	66.4

#### Table 2 Trends of Intra-Regional Trade in East Asia and Europe

(in percentages of total trade)

Note:

This table is Table 4 in Lee, Park and Shin (2002). Intra-regional trade is measured by the share of an economy's trade with the rest of the economies that belong to the same region, in total trade.

Source: International Monetary Fund, Direction of Trade Statistics.

<sup>8</sup> Free capital mobility was not allowed until the last stage of currency unification in Europe. This is because the conflict between exchange rate stability and the active use of monetary policy was reconciled through internal and external financial repression. As Europe has liberalised its capital account, such a formal coordinated (fixed but adjustable) exchange rate mechanism was greatly endangered by volatile capital movements. When capital controls were lifted, currency union was preferred to maintaining national monetary policies. See Wyplosz (2001) for more details. Financial claims are all denominated in US dollars and the bulk of foreign lending and borrowing is intermediated through international financial markets in New York and London. As far as finance is concerned, therefore, gains from adopting a common currency in terms of lower transaction costs and foreign exchange risks may not be as large as they could be when regional financial markets are integrated. In particular, Singapore and Hong Kong, as regional financial centres, could lose a substantial portion of their income from the financial services industry.

As for institutional and political constraints on further expansion of the CMI, the most serious one has been that the thirteen countries have failed to articulate the ultimate objectives of the CMI arrangement. The participating countries themselves are still unclear about whether the CMI is going to be fostered as a regional liquidity support programme or as a building block for a full-fledged regional monetary system in East Asia. If bilateral swap arrangements are activated collectively and supported by a surveillance system, then they constitute a *de facto* regional monetary fund. The CMI could then be used as the base on which an elaborate system of financial cooperation and policy coordination is built by following in the footsteps of the European monetary integration.<sup>9</sup> At this stage of development, many countries in East Asia are not prepared to accept the idea of or may feel uneasy about restructuring the CMI into a forerunner of the AMF.

A second institutional constraint is related to the need to coordinate the activities of the CMI with other regional arrangements such as the Manila Framework supported by the US, Australia, and New Zealand. Most of the CMI countries also participate in the Manila Framework and APEC. At some point in the future, the leaders of the ASEAN+3 countries may have to decide on the mode of cooperation and division of labour in promoting regional growth and stability between these institutions and the CMI. All thirteen countries have been engaged in

<sup>&</sup>lt;sup>9</sup> From the theoretical point of the neo-functionalists, initial steps toward integration trigger self-sustaining economic and political dynamics leading to further cooperation. Economic interactions create spillovers or externalities that need to be coordinated by governments involved. Such economic policy coordination at the regional level can be seen as an inevitable response to the increased economic interactions within the region. Once the integration process starts, spillovers deepens and widens integration by working through interest group pressures, public opinion, elite socialisation or other domestic actors and process (George, 1985).

policy reviews and dialogues through the various APEC meetings and the Manila Framework. Unless the CMI is developed into a credible financing mechanism by increasing swap amounts, it will take on a role similar to other regional economic forums. The coherence of the group will then be weakened, as questions are raised as to whether the thirteen countries constitute an appropriate grouping for a regional financing arrangement in East Asia.

A third hindering factor is that the fear of another round of financial crisis has receded with a recovery that has been faster than predicted on the basis of previous episodes of crises. With the speedy recovery, the ASEAN+3 countries have become less interested in enlarging and institutionalising the CMI operations. Instead, their focus has recently shifted to creating free trade areas in East Asia (see Table 3). The ASEAN states have already agreed to establish a free trade area among them. Japan has concluded a free trade agreement with Singapore and proposed negotiations on a similar agreement with Korea. China has announced its interest in negotiating free trade with the ASEAN and other neighbouring countries.

The free trade movement is undoubtedly a desirable development, and the CMI could facilitate further liberalisation of trade by stabilising bilateral exchange rates of regional currencies and minimising the disruptive effects of financial market turbulence. This advantage suggests that the ASEAN+3 countries may have an incentive to broaden the scope of the CMI in parallel with negotiations on establishing free trade areas in the region. In reality, however, it appears that free trade discussions have rather distracted many East Asian countries from their CMI negotiations.

Finally, there is the leadership issue that defies an easy solution. If the thirteen countries have a more ambitious goal of developing a collective exchange rate mechanism similar to the ERM in Europe with the long-term objective of adopting a common currency, they will have to increase the number and amounts of the BSAs. As the European experience shows, such an extension requires leadership that can foster coherence among the thirteen countries by mediating between the divergent interests of the members.

China and Japan are expected to provide leadership in forging regional consensus for expanding and consolidating the BSAs as a regional institution, but they have not been able to agree on a number of operational issues including the surveillance mechanism. Except for Japan, no other potential swap lenders including China

	Year	Participants and Status			
FTA in Force					
ASEAN Free Trade Area (AFTA)	1992	10 ASEAN members			
Australia-New Zealand Closer Economic Relations Trade Agreement (CER)	1983	Australia, New Zealand			
Singapore-New Zealand FTA	2001	Effective in January			
Japan-Singapore Economic Partnership Agreement (JSEPA)	2002	Effective in November			
Singapore-EFTA (European Free Trade Association) FTA	2002	Signed in June and effective in January 2003			
Korea-Chile FTA	2003	Signed in February			
Agreements Being Negotiated, Studied, or Considered					
East Asia Free Trade Area (EAFTA)	2000	Proposed at the ASEAN+3 summit meeting			
China-Japan-Korea FTA	2000	Chinese Premier Zhu Rongji proposed during the ASEAN+3 summit meeting			
ASEAN-China Free Trade Area (ACFTA)	2001	Realisation by 2010 (Framework Agreement signed in 2002)			
Japan-ASEAN Closer Economic Partnership	2002	Realisation within ten years agreed to at an ASEAN-Japan Summit meeting			
ASEAN-India Regional Trade and Investment Agreement	2002	Consideration of an agreement agreed to at the ASEAN-India summit meeting			
Bilateral FTA Under Consideration					
China		Hong Kong SAR, Macao SAR			
Japan		Mexico, Philippines, Korea, Thailand			
Korea		Japan, Mexico, Thailand, ASEAN			
Singapore		Australia, Canada, Mexico, United States			
Thailand		Australia, India, Japan			

#### Table 3 Free Trade Agreements in East Asia

Source: Wei (2003) and various other sources.

are prepared to increase the amounts of their bilateral swaps with other contracting parties. Japan could increase its swap amounts with the ASEAN states and Korea (under the presumption that China will not borrow from Japan) to make the CMI a more credible financing scheme. However, unless Japanese authorities receive some sort of assurance that their short-term lending will be repaid, they are not likely to lead an expansion and institutionalisation of the CMI. As a minimum condition for expansion of the CMI, Japan would demand the creation of an effective surveillance mechanism for the region in which it can exercise influence commensurate with its financial contribution. However, China may feel that it cannot play the second fiddle to Japan in any regional organisation in East Asia. This concern appears to be the most serious roadblock to further development of the CMI.

China and Japan have different interests and hence different strategies for economic integration in East Asia. As far as China is concerned, economic integration with the ASEAN ten members, South Asian and Central Asian countries may be more important both economically and geo-politically than financial cooperation or free trade with either Japan or South Korea. While China is a super military power in the world, it is still a developing economy with a huge gap to narrow in terms of technological and industrial sophistication vis-à-vis Japan. Although China has been growing rapidly, it has a long way to go before catching up with Japan. These differences in the economic and military status of the two countries suggest that, even if they manage to reconcile their troubled memories of the past, China and Japan may find it difficult to work together as equal partners for regional integration in East Asia.<sup>10</sup>

<sup>&</sup>lt;sup>10</sup> France and Germany also had a wartime legacy. Although de Gaulle's nationalism was generally popular within the country, he also appreciated that membership of the common market would benefit France economically. However, de Gaulle remained implacably opposed to any increase in the powers of the European Commission, or to any other increase in supranationalism. He showed just how opposed he was in 1965, when he precipitated the most dramatic crisis in the history of the European Community (George, 1985). It was German Chancellor Helmut Schmidt and French President Giscard d'Estaing that accelerated the stalled integration process at the end of the 1970s. The joint initiative of Chancellor Helmut Kohl and President François Mitterand resulted in a great leap towards EMU in the beginning of the 1990s. The Franco-German alliance formed the core for the integration process in Europe, as it was the political will of these two countries that motivated further integration.

China borders Russia and many of the South Asian and Central Asian countries in addition to several ASEAN members. Therefore, it is natural for China to seek expansion and deepening of its trade and financial relations with those neighbouring countries. In fact, for this reason, China has been courting ASEAN for a free trade agreement and joined in November 2001 the Bangkok agreement on a free trade area that includes Russia and the South Asian countries. China has also taken a leading role in establishing the Shanghai Cooperation Organisation, a cooperative arrangement between Russia, Kazakhstan, Kyrgyzstan, Tajikistan, and Uzbekistan and China.<sup>11</sup>

In contrast, Japan has not been able to articulate its strategic interests in East Asia. While Japan has been at the forefront in supporting greater economic cooperation among the East Asian countries, its perspective on the geographical contiguity of East Asia has not been altogether clear. Japan has been promoting integration among the "ASEAN+5", but which are the two countries added to ASEAN+3? At times, the five countries are stated as China, Japan, Korea, Australia, and New Zealand, and at other times Taiwan and Hong Kong replace Australia and New Zealand.

There is also the suspicion that Japan is not interested in free trade and financial arrangements *per se* in East Asia for purely economic reasons. Instead, Japan is engaged in the discussion of those regional arrangements with other East Asian countries to maintain its leadership role as the region's largest economy by checking and balancing China's expansion. Many analysts believe that Japan's active involvement in regional economic integration is therefore motivated by its desire to maintain its traditional pole position.<sup>12</sup> On top of this suspicion, Japan is perceived to be a country insensitive to and unwilling to resolve wartime legacies and disputes on historical and territorial claims. Japan has also been

<sup>&</sup>lt;sup>11</sup> In June 2001, the presidents of five countries signed the Declaration of the Shanghai Cooperation Organisation (SCO). The SCO aims at strengthening mutual trust and friendly relations among member states, encouraging their further effective cooperation in politics, economy, science and technology, culture, education, energy, transportation, environmental protection and other fields, jointly ensuring regional peace, security and stability, and creating a new international political and economic order.

<sup>&</sup>lt;sup>12</sup> See David Wall, "Koizumi Trade Pitch Nests", In: *Japan Times*, April 21, 2002.

gripped with a decade long recession and unability to restructure its economy.<sup>13</sup> These developments combined with its lack of a strategy for East Asian development seem to undermine Japan's capacity to pull East Asian countries together for regional cooperation and integration.

#### **4** Future Prospects

What are then the likely courses of development of the CMI? How would regional financial integration proceed in East Asia? One possible scenario is that China and Japan will come to realise that despite the differences in their strategies, the consolidation of the CMI group would serve their interests. This realisation could soften their positions to compromise on an institutional setting and augmentation of the existing BSAs. For instance, China may accept Japan's demand for *de facto* control over monitoring and surveillance in return for Japan's pledge for a substantial increase in financial assistance in the form of one-way swaps and ODA to ASEAN members. China could agree to this scheme, if it is confident about concluding a free trade agreement with the ASEAN members in the near future. China's free trade pact with ASEAN could circumscribe Japan's influence on ASEAN affairs even if Japan is a major provider of financial resources to the region.<sup>14</sup>

Another scenario focuses on the possibility of China assuming a more aggressive leadership role in regional integration. In view of the uncertain prospects of the Japanese economy, China could emerge as the region's engine of growth over the longer term if it sustains its growth. Given the envisaged leadership role, China may choose to negotiate both the expansions of the BSAs and a free trade

<sup>&</sup>lt;sup>13</sup> Uncertain economic prospects may make Japan unlikely to be the driver in the region's integration movement as it was in the past. China is emerging both as a strong competitor and as a promising market.

<sup>&</sup>lt;sup>14</sup> The dramatic declines in Japanese equity and commercial prices in the early 1990s put strong downward pressures on the capital positions of Japanese banks. In responding to these domestic shocks, Japanese banks have reduced overseas lending (Peek and Rosengren, 1998; 2000). In particular, Japanese banks repatriated most of their funds during the crisis; Japanese banks withdrew \$38 billion in the six quarters from mid-June 1997 to the end of December 1998 (de Brouwer, 2001).

pact with ASEAN. In this case, the original CMI would become "ASEAN+1" in the sense that Japan could play the second fiddle. Realising that financial integration is an integral part of a successful free trade area, China may indeed seriously consider this option. However, without Japan, ASEAN+1 will not be a viable arrangement for a regional financing scheme simply because China is hardly in a position to commit itself to financing the balance of payments deficits of all ASEAN member states. It is also questionable whether ASEAN will join any regional financial arrangement in which China is the dominant member.

A third scenario is the enlargement of the CMI to include Australia and New Zealand and possibly India from South Asia. This is the route favoured by Japan in the sense that Japan would find it easier to deal with China when there are more countries supporting its strategy. However, many members of ASEAN+3 believe that at this stage forming a critical mass of the CMI should precede any enlargement discussion.<sup>15</sup> Since the enlargement is not likely to increase substantially the availability of short-term financing, most members of ASEAN+3 would not take the third scenario seriously.

Perhaps the most realistic scenario is that the countries participating in the CMI will muddle through, continuously discussing modalities of policy dialogue, the types of the surveillance system the CMI needs, and also augmentation of swap amounts without making any substantial progress. However, a possible breakthrough may come over the next few years as the economic consequences of European monetary unification become better understood. The enlargement of the EU in 2004, when eight Central and East European countries are expected to be admitted, will also have a large impact on the thinking of East Asian policymakers on regionalisation in East Asia. If the members of the European monetary union make a smooth adjustment to the single currency and the EU enlargement proceeds as planned, then these developments will give a strong impetus to East Asian integration.

<sup>&</sup>lt;sup>15</sup> Kumar (2002) asserts that the five blocs of the Asian regional economy, including Japan, ASEAN, China, India and Korea (JACIK), can form an effective skeleton in terms of management and coherence for the creation of an Asian Economic Community. This approach may be called as "ASEAN+4" in our terminology.

#### References

- Bergsten, C. Fred (2000), "Toward a Tripartite World", In: *The Economist*, No. 356, July 15, pp. 23-25.
- Baek, Seung Gwan, and Chi-Young Song (2002), "Is Currency Union a Feasible Option in East Asia?", In: H. Choo and Y. Wang (eds.), *Currency Union in East Asia*, Korea Institute for International Economic Policy, Seoul.
- Choe, Jong-Il (2001), "An Impact of Economic Integration through Trade: On Business Cycles for Ten East Asian Countries", In: *Journal of Asian Economics*, Vol. 12, pp. 569-586.
- de Brouwer, Gordon J. (2001), "East Asian Financial Markets: Current Status and Future Development", In: Y.H. Kim and Y. Wang (eds.), *Regional Financial Arrangements in East Asia*, Korea Institute for International Economic Policy, Seoul.
- Dieter, Herbert (2000), "Monetary Regionalism: Regional Integration Without Financial Crises", CSGR Working Paper No. 52/00, Center for the Study of Globalization and Regionalization, University of Warwick, May.
- Eichengreen, Barry and Tamin Bayoumi (1999), "Is Asia an Optimum Currency Area? Can It Become One? Regional, Global and Historical Perspectives on Asian Monetary Relations", In: Stefan Collinon, Jean Pisani-Ferry and Yung Chul Park (eds.), *Exchange Rate Policies in Emerging Asian Countries*, Routledge Studies in the Growth Economies of Asia, Routledge, London and New York.
- George, Stephen (1985), *Politics and Policy in the European Community*, Clarendon Press, Oxford.
- Henning, Randall C. (2002), *East Asian Financial Cooperation*, Policy Analyses in International Economics, Institute for International Economics, September.
- Kumar, Nagesh (2002), "Towards an Asian Economic Community: Vision of Closer Economic Cooperation in Asia", mimeo, Research and Information System for the Non-Aligned and other Developing Countries, New Delhi.
- Lee, Jong-Wha, Yung Chul Park and Kwanho Shin (2002), "A Currency Union in East Asia", mimeo, Korea University.
- Loayza, N., H. Lopez, and A. Ubide (2001), "Co-movements and Sectoral Interdependence: Evidence for Latin America, East Asia and Europe", IMF Staff Papers 48, IMF, Washington D.C., pp. 367-395.

- Manzano, George (2001), "Is There Any Value-Added in the ASEAN Surveillance Process?", In: *ASEAN Economic Bulletin*, Vol. 18, No. 1, April, pp. 94-102.
- Park, Yung Chul (2002), "Prospects for Financial Integration and Exchange Rate Policy Cooperation in East Asia", ADB Institute Research Paper No. 48, Asian Development Bank Institute, Tokyo.

— and Yunjong Wang (2002), "Can East Asia Emulate European Economic Integration?", KIEP Discussion Paper No 02-09, Korea Institute for International Economic Policy, Seoul.

- Peek, J., and E.S. Rosengren (2000), "Collateral Damage: Effects of the Japanese Bank Crisis on Real Activity in the United States", In: *American Economic Review*, Vol. 90, pp. 30-45.
- —— and E.S. Rosengren (1998), "Japanese Banking Problems: Implications for South-East Asia", paper prepared for the Second Annual Conference of the Central Bank of Chile on "Banking, Financial Integration, and Macroeconomic Stability", Santiago de Chile, 3-4 September.
- Scalapino, Robert A. (1991), "The United States and Asia: Future Prospects", In: *Foreign Affairs*, Vol. 70, No. 5, pp. 19-40.
- Shin, Kwanho and Yunjong Wang (2003), "Trade Integration and Business Cycle Synchronization in East Asia", Asian Economic Papers, forthcoming.
- Wang, Yunjong and Wing Thye Woo (2002), "A Study on Information Exchange, Surveillance Systems, and Regional Institutions in East Asia", Report to the Asian Development Bank in fulfillment of TA 6000 REG/Study on Monetary and Financial Cooperation in East Asia (JSF).

— and Deok Ryong Yoon (2002), "Searching for a Better Regional Surveillance Mechanism in East Asia", KIEP Discussion Paper 02-01(a part of the ASEM-Kobe Research Project), Korea Institute for International Economic Policy, Seoul.

— (2001), "Instruments and Techniques for Financial Cooperation", paper presented at the conference on "Regional Financial Arrangements in East Asia", jointly organised by the Japanese Ministry of Finance and Australian National University, Canberra, 12-13 November 2001.

— (2000), "The Asian Financial Crisis and Its Aftermath: Do We Need a Regional Financial Arrangement?", In: *ASEAN Economic Bulletin*, Vol. 17, No. 2, pp. 205-217.

- Wei, Li (2003), "A Road to Common Prosperity: Examination of an FTA between India and China", paper presented at the international conference on "Building a New Asia: Towards an Asian Economic Community", New Delhi, 10-11 March.
- Wyplosz, C. (2001), "Regional Exchange Rate Arrangements: Some Lessons from Europe", In: Jan Joost Teunissen (ed.), A Regional Approach to Financial Crisis Prevention: Lessons from Europe and Initiatives in Asia, Latin America and Africa, FONDAD, The Hague.

# 7

## Comment on Yung Chul Park and Yunjong Wang

Charles Adams

L et me begin by thanking the conference organisers – the Forum On Debt and Development (FONDAD), the Korea Institute for International Economic Policy (KIEP) and the School of International and Area Studies (SIAS) – for inviting me to participate in the conference, and comment on this interesting paper by Yung Chul Park and Yunjong Wang (henceforth Park and Wang). The theme chosen for the conference is a very timely and important one and the organisers are to be congratulated for assembling an impressive range of papers and speakers. Park and Wang's paper fits nicely into the overall theme of the conference and, at the same time, addresses important issues faced by regional policymakers as they seek to deepen and strengthen regional cooperation.

Park and Wang's main contribution is to take stock of current efforts to deepen financial cooperation in the region – the ASEAN+3 group and the Chiang Mai Initiative are the most visible manifestations of these efforts – and then assess the prospects for further progress. In the process, the authors outline key features of the Chiang Mai Initiative, consider key outstanding issues of both a technical and political nature, and discuss possible roadmaps for the future. Overall, the assessment reached is a fairly negative one: prospects for significant further progress are judged to be low. In fact, the authors characterise the most likely scenario as one in which participants in the Chiang Mai Initiative continue to "muddle" through, without coming to substantive agreement on important outstanding issues such as the creation of a formal regional monitoring and surveillance mechanism, modifying the swap arrangements etc. While the reasons for this somewhat negative prognosis are not entirely clear, one is left with the impression – reading between the lines – that the main problems are judged to be a lack of leadership on the part of the largest participants and an associated failure to agree on the ultimate objectives of the exercise (Achievement of monetary union? Stable intra-regional exchange rates? Regional free trade?).

As an outsider who has not been involved in the Chiang Mai Initiative, it is obviously difficult to comment on many of the specific issues that have arisen in the efforts to implement and build upon the initiative. The authors' analysis, however, seems consistent with many other researchers who have written about the initiative and with what one hears "in the corridors" from participants in the process. Having said this, however, one of the main reasons the authors seem unusually negative about the prospects for Chiang Mai is that they are clearly quite ambitious in their assessment of what could and should be accomplished in a short period of time. A contrarian view – maybe this is a case of a glass being half full rather than half empty - would be that deepening financial cooperation will necessarily take time, and that what is important is that efforts are being made, mechanisms for regional dialogue are being developed, the appropriate financial arrangements for the region are being actively discussed, and countries are starting to undertake peer reviews etc. of their policies. If, like me, one subscribes to this more optimistic view then it may not matter too much if reaching agreement on the "big" issues (such as the ultimate objective of the exercise) or on some of the "smaller" issues (such as the precise regional surveillance mechanism) is taking a long time. What matters is that key issues and options are being discussed.

Against this background, let me say something about one of the key arguments of the paper, namely, that further progress on the Chiang Mai Initiative will depend importantly on the two largest participants, Japan and China, reaching an understanding of their respective roles, and there being agreement on the ultimate objectives of the exercise. While I concur with this assessment, the two main scenarios the authors lay out for the possible roles of these countries did not seem to me to be very plausible. Let me briefly summarise the two main scenarios. According to one scenario, Japan and China (implicitly or explicitly) agree on a division of labour in the area of regional cooperation. In particular, Japan would take the lead in providing regional financing and setting up a regional surveillance mechanism while China would focus on negotiating bilateral and regional free trade agreements in the region, notably with ASEAN and its members. According to the second scenario, China would assume a dominant role in both regional financing arrangements and the promotion of regional trade, with Japan playing a much less important role in both areas.

As I am not a political scientist, I would not want to try to delve very much into the political assessments underlying the two scenarios. I would simply note, however, that neither seems very likely on economic grounds since both go to "corner points" with regard to the role of either Japan or China, and neither is therefore in the cooperative spirit of Chiang Mai. Moreover, both scenarios seem to imply that the key issue is how some – or all – of the "plus 3 group" of countries will interact with ASEAN as a group rather than among themselves. It is too early, in my view, to exclude the possibility of mutually beneficial cooperation within ASEAN+3, with both Japan, Korea and China, together with ASEAN, agreeing on mutually beneficial arrangements. Nevertheless, casting the issues in the stark terms used by the authors is useful in underscoring that Japan and China may have different interests regarding the future of regional cooperation and that finding common ground will be important in determining the evolution of Chiang Mai.

Thinking about the issues in these terms also raises the possibility that the factors driving regional cooperation in the future may be different than in the past. In particular, there is the possibility that the overall emphasis may shift in the next few years towards supporting the promotion of free trade and investment in the region and developing the financing and other arrangements that best support such an objective. Indeed, in looking back at the factors that were initially important in the Chiang Mai Initiative, the "driver" that clearly stands out is the very negative experience in the Asian financial crisis, along with the view that self-help mechanisms are needed at the regional level to help avoid and forestall future crises. Accordingly, the emphasis in ASEAN+3 initially was on setting up a regional financing mechanism and strengthening

regional monitoring and surveillance, with crisis prevention and management as the ultimate objectives. (Relatedly, of course, there has also been interest in promoting local and regional bond markets with a view to helping avoid the currency and maturity mismatches that contributed importantly to the severity of the 1997-98 crisis. Regional solutions have also been supported by important reforms at the country level to help strengthen financial systems and lower external vulnerability).

While crisis prevention will no doubt remain important as the Chiang Mai Initiative evolves, the emergence of China as a major player in the global trading system and the associated growth of intra-regional trade may also tend to shift the emphasis in the initiative. In particular, more attention might in the future be paid to promoting trade and investment in the region and to seeking to minimise disruptions to inter-regional trade that could result from financial problems or instability in one country spilling over to other countries. If I interpret Yung Chul Park's oral remarks correctly, this is what he has in mind when he refers to "new" issues associated with China's emerging role in the region, that might lead either to a change in focus in Chiang Mai or to a strengthening of the case for certain types of financial cooperation.

If issues associated with China's emergence do become more important, it certainly seems plausible that there might be a broadening of the agenda addressed by ASEAN+3 and consideration of complementary – and appropriately sequenced – means to deepen both financial and trade cooperation. If this occurs, I would conjecture that the efforts would continue to be outward rather than inward looking. This would be because the participants in ASEAN+3 have a strong interest in Asia's close integration with the global economy and in helping manage the opportunities and challenges associated with globalisation. In short, regional solutions will likely continue to be seen as a means to strengthen Asia's ability to participate in - and continue to benefit - from globalisation rather than being inward-looking in character. This said, however, it will be important that current efforts to negotiate bilateral and regional free trade agreements - within, among, and beyond ASEAN+3 members - not only be WTO consistent but also do not lead to a patchwork of agreements with different "rules of origin etc.". The reason for this is that a lot of "different" agreements might be difficult to put together into a region-wide free trade

agreement, should there be eventually interest in such an arrangement.

Finally, let me conclude by saying again that I found Park-Wang's paper very stimulating in thinking about the future of the Chiang Mai Initiative. Although I share some of Park-Wang's concerns about the apparent lack of progress in deepening financial cooperation, my sense is that the process will take time and continue to evolve. Far better, in my view, is to spend the necessary time to reach consensus on the "means" and "objectives" of financial cooperation – and the roles of different participants – than to enter into any substantive agreements that may not have widespread support, and might therefore lack credibility or staying power. In addition, it is possible that China's growing importance in the region and the global economy could provide a new impetus for – and lead eventually to broader and stronger – approaches to regional cooperation that more explicitly link the financial and trade dimensions.

# 8

## Global Capital Flows and the Position of China: Structural and Institutional Factors and their Implications

Young Rok Cheong and Geng Xiao

Global capital flows into developing economies have been driven by two fundamental factors: profit opportunities in the emerging economic frontiers and the physical and institutional barriers to international capital mobility. Advancement in transportation and communication technology has greatly reduced the physical costs of cross-border mobility of capital and goods while the spread of global development knowledge has also led to new profit opportunities in less developed economies, such as China.

Since the early 1980s, China has emerged as a major global development frontier, following its open-door policies and economic reforms initiated by Deng Xiaoping. Since that time, the barriers to foreign trade and investment in China have declined steadily, leading to China's accession to the World Trade Organisation in late 2001 – after one-and-a-half decades of tough negotiations. By the end of 2002, only a year after joining the WTO, China overtook the US in FDI inflows, becoming the most attractive FDI destination in the world, receiving \$52.7 billion in FDI.

China's recent dramatic achievement seems to suggest that today's global economy is unprecedented in terms of openness and the amount of foreign capital flows into developing countries.

113

Unfortunately, this optimistic impression is not confirmed by facts. Foreign capital flows into developing countries today are far below historical record achieved before World War I. The gross value of foreign capital stock in developing countries increased from 8.6% of their GDP in 1870 to a peak of 32.4% in 1914. This ratio dropped to 4.4% in 1950 due to the interruption to the global trade and investment during the war and by 1973, recovered only to 10.9% and 21.7% by 1998 (Maddison 2001, p. 28). Hence, in spite of technological advances during the last century, the world is actually less open for capital flows to less developed countries than one hundred years ago.

Capital flows among developed countries are much freer than between developed and developing countries because of better protection of property rights and less capital control in the developed economies. High capital mobility among developed economies has led to a dramatic change in the holding of net foreign assets among developed countries during the last decade. For example, from 1989 to 1998, Japan's holding of net foreign assets increased from \$294 to \$1.153 billion while the US obligation in net foreign liabilities jumped from \$49 to \$1.537 billion (Maddison 2001, p. 137). Clearly, Japan has exported a substantial amount of capital to the US in search of better risk-adjusted return and in preparation for its aging population, even while the policy environment in Japan, such as the volatility of the exchange rate and the secular appreciation of the yen, is unfavourable to Japanese investment in foreign assets.

In the last decade, foreign direct investment in China increased from about \$3 billion in 1990 to \$52.7 billion in 2002. As impressive as this is, on a per capita basis, China's FDI inflow in 2001 was still below the average of FDI inflows to the rest of developing countries. Per capita FDI inflows in 2001 was only \$37 for China, compared to an average of \$120 for the world and \$42 for all the rest of developing countries.

Foreign invested enterprises in China have contributed to more than half of China's exports. China has been generating current account surpluses for the last nine years. Since current account surplus simply means net savings or net export of capital, China is taking in FDI on the one hand and exporting capital to capital-rich economies like the United States on the other hand. How can we reconcile these seemingly inconsistent patterns of capital flows? Is China attracting too much FDI? What happened to China's investment in US bonds? What is the impact of foreign portfolio investment on the stability of the Chinese financial markets and other Asian regional markets? Is China generating global deflation? How serious is China's impact on its competitors? Is China saving too much? Should China revaluate its currency? Is China becoming a growth engine for the world? These are questions we examine in this chapter.

The next section presents the basic facts on the patterns of global capital flows and the position of China. Section 2 explains the uniqueness of China's economic conditions in attracting capital inflows. Section 3 assesses the impact of China's rapid integration into the global economy on the rest of the world. Throughout our analysis, we emphasise structural and institutional factors such as the gaps in development, population structure, and financial and legal institutions and their effects on the patterns of global capital flows and global economic development. Section 4 concludes.

#### 1 Global Capital Flows and the Position of China

In this section, we first summarise the pattern of global trade and foreign direct investment based on the extensive information from the WTO and the World Investment Report of UNCTAD. Then we review global portfolio investment using detailed IMF and US Treasury surveys. Our purpose here is to provide a concise and comparative background to determine China's position in the global capital market.

#### US Trade Deficits with China and the Asian Region

Global merchandise trade increased from \$124 billion in 1948 to 12.7 trillion in 2000, an increase of more than 100 times over a period of half-century. Western Europe and North America have maintained about 40% and 20% of the global trade respectively throughout this period. Asia's share has risen from about 14% to about 24% at the costs of Latin America, Africa, and other regions.

During the last two decades from 1982 to 2001, the growth rate for global merchandise trade varied from -5.8% in 1982 to 19.5% in 1995, averaging about 6.1% a year. The global trade in commercial

services was relatively stable at a level of about 23% of the merchandise trade (see Table 1.1<sup>1</sup>).

In 2000, Asia's share of global exports (at 26.7%) is 3.9% higher than its share of imports (at 22.8%) while North America's share of exports (at 17.1%) is 6.1% lower than its share of imports (at 23.2%) due to exceptional export performance in Japan, Asia's newly industrialising trading economies and China. From 1948 to 2000, Japan increased its share of global export from 0.4% to 7.7% and six Asian traders (see Table 1.2) increased their share from 3% to 10.5%. In the last quarter of the 20th century, China increased its share of global exports from 1.2% in 1983 to 4.3% in 2001, with a strong momentum for further gains of market share in the future (see Table 1.2).

In 2001, the US current account deficit (net capital import) reached \$393.4 billion. On the other side, current account surplus (net capital export) was \$87.8 billion for Japan, \$57.1 billion for the six Asian traders, \$17.4 billion for China, and \$39.6 for transition economies (see Table 4.7). Except for Japan, many countries with a current account surplus (net capital exporters) are not capital-rich economies. In 2000, according to the IMF, the US absorbed 64% of global net capital exports (i.e. the sum of the current account surpluses of the rest of the world).

Who is financing the net capital import to the United States? Table 4.10 shows the exports, imports and balance of goods account for the US in 2002. The US goods deficit, which is the major part of its current account deficit, is as high as \$484 billion. The US goods account deficit is financed by and distributed among the rest of the world: 18% by North America, 18% by Western Europe, 14.5% by Japan, and 21.3% by China. The importance of China in US trade deficits is at the centre of the globalisation debate. Is China taking away jobs from American workers or financing American consumers? Is China generating global deflation? Is China saving too much? Should China revaluate its currency? We discuss these questions later.

Due to rapid regional integration of production through the socalled supply chain management, China's strong export performance is intimately related to the export activities of its neighbours. This modern management model, pioneered by Hong Kong

<sup>&</sup>lt;sup>1</sup> All tables and graphs are at the end of this chapter.

entrepreneurs, separates the whole value-added chain into different parts and searches for the best deals in distribution and production among global providers of production and services. This innovation makes our traditional concept of "made-in-China" or "made-in-Japan" much less relevant as many parts of final goods and services are produced or provided by countries other than the one where final goods are exported.

Table 4.9 and Graph 5.2 show the rising US current account deficits with the greater China, including Hong Kong and Taiwan. It is clear that the part of trade deficits attributable to Hong Kong and Taiwan are either declining or stabilising while the part due to China is rising rapidly. This is largely because the production of final goods has been rapidly relocated to China from Hong Kong, Taiwan and other Asian economies over the last decade. But the key components or high value-added parts of the supply chain are still kept in the more developed Asian economies. If this part of the contribution to the production of final goods is excluded, China's own value added in exports to the US would be very small. This point needs to be remembered as we attempt to assess China's impact on the world. About two-thirds of China's FDI is coming from its neighbours such as Hong Kong, Macao, Taiwan, Japan, Singapore and Korea, in part for the purpose of assembling final products to be exported to North America and Europe.

#### Steady FDI Flows into China

Foreign direct investment is much more volatile than foreign trade, but it is becoming increasingly important in recent decades for both developed and developing countries. The growth rate for global FDI fluctuated widely from 56.7% in 1999 to -50.7% in 2001, averaging about 16.3% in the last two decades. As a result of rapid FDI growth, global FDI flows as a percentage of global merchandise trade increased from about 1.5% in the early 1980s to between 6% and 12% in the last five years. But the scale of global FDI flows is still much smaller than the global service trade, which is about 23% of merchandise trade. At its peak in 2000, global FDI reached \$1.5 trillion, or about 15% of the United States' GDP (Table 1.1).

Most of global FDI, especially FDI among developed countries, is through mergers and acquisitions (M&A) rather than green-field investment. In 2001, M&A amounted to as much as 80% of global

FDI. Among all the M&A in 2001, 83.5% was conducted in the developed countries, 31.1% in the US alone, and only 5.8% in Asia and the Pacific region.

Developed countries are key hosts of FDI with their share of global FDI at levels ranging from 60% to 80%. At the regional level, North America has been expanding its market share at the costs of European countries although the latter have been picking up in the late 1990.

Graph 5.1 shows the trend since 1979 of FDI inflows to US, China, Asia and the Pacific region, and all developing countries excluding China. The US has been dominating the global FDI inflows with its share ranging from 11.2% or \$19 billion in 1992 to 26% or \$283 billion in 1999. Asia and the Pacific region have increased their market share from 15.2% in 1991 to 24.3% in 1996, but the Asian financial crisis depressed their share to only 9% in 2000.

China's share of global FDI increased from a low base of 1.7% in 1990 to a peak of 13% in 1994. After 1994, China's share of global FDI declined steadily to only 2.7% in 2000 largely due to massive M&A activities in the developed economies during the tech bubble. After the burst of the tech bubble, global FDI dropped 50% in 2001 but China's FDI was growing steadily, contributing to a recovery of China's share in global FDI to 6.4%, which is consistent with its trade expansion to 4.3% of the global export by 2001.

FDI into China has exceeded \$40 billion since 1996 and has been growing steadily every year since 1990. This puts pressure on other developing countries, especially its Asian neighbours. As shown in Table 1.3, the Asia-7, including India, Indonesia, Malaysia, Philippines, Republic of Korea, Singapore, and Thailand, with more population than China, only had \$33 billion FDI inflows at their peak year of 1997. After the Asian financial crisis in 1997-1998, the Asia-7's FDI inflows declined dramatically to only \$18 billion by 2001. The Asian financial crisis however did not slow FDI flows into the developing economies as a whole. FDI into developing economies excluding China recorded a steady growth from \$34 billion in 1990 to \$147 billion in 1997, and peaked at \$197 billion in 2000, and then fell to \$158 billion in 2001.

In 2001, per capita FDI inflows are \$120 for the world, \$420 for the developed economies, \$42 for the developing economies excluding China, \$37 for China, and only \$12 for the Asia-7.
Apparently China is winning the competition for FDI inflows and its neighbours are very much concerned about this trend.

Table 1.6 shows inward FDI stocks for selected regions and countries over the period from 1980 to 2001. Global FDI stock increased from \$636 billion in 1980 to \$6258 billion in 2000, an increase of almost ten times. During the same period, the world trade volume only increased about three-folds from \$4 trillion in 1980 to \$12.5 trillion in 2000. This clearly shows the increasing importance of FDI in the world economy and the expanding scope and depth of globalisation.

However, the access to foreign capital is unequal with 5 billion out of a 6.1 billion world population in the developing countries receiving only \$2.1 trillion out of \$6.8 trillion in the FDI stock by 2001. In 2001, per capita FDI stock is \$1,118 for the world, \$3,763 for the developed economies, \$478 for all developing economies excluding China, \$309 for China, and only \$220 for the Asia-7.

The developed economies provided most of the global FDI stock but its share is declining from 95.8% in 1980 to 87.8% in 2001. In the last decade, Hong Kong emerged as a major financial centre for facilitating capital flows into China. Hong Kong's outward FDI stock increased from \$2.3 billion in 1985 to \$375 billion in 2001, exceeding Japan's \$300 billion. In 2001, Hong Kong captured 5.7% of global FDI outward stock, compared with only 4.6% for Japan. A significant part of Hong Kong's outward FDI into China is "roundtripping" Chinese capital, perhaps as much as one-quarter (World Bank, 2002, p. 41).

The pattern of global FDI flows raises a few core questions. Is China attracting too much FDI? Is China hurting its Asian neighbours and competitors? These questions can only be discussed meaningfully after examining many special development conditions in China.

#### Two-Way Flows in Cross-Border Portfolio Investment

Based on the recent *Global Portfolio Investment Survey* by the IMF (Table 2.1), the derived portfolio investment liabilities for the world in 2001 are \$12.5 trillion, about twice the amount of the global FDI stock and almost equal to the sum of global merchandise exports and imports. The IMF survey is so far the best estimate on the stock of global portfolio investment. Compared to IMF's last survey for

1997, the global portfolio investment stock doubled in only four years. The top three targets for portfolio investment stock are very stable, including the US, UK and Germany, sharing 24.5%, 10%, and 9.2% of the global total respectively. Japan fell from the fourth place in 1997 to the sixth in 2001, now behind France and the Netherlands. Japan's share dropped from 6.5% in 1997 to 4.2% in 2001, reflecting its weakening economy (see Table 2.1).

China made little progress in attracting foreign portfolio investment during 1997 to 2001. The derived amount of foreign portfolio investment in China increased slightly from \$19.3 to \$20.1, reflecting its stagnant "B shares" market, which is a tiny experimental stock market designed for foreign investors with share prices quoted and traded in foreign exchange.

In March 2001, China opened its "B share" market to domestic residents with foreign exchange savings. This opening caused a brief surge in prices and many foreign investors took profits and dumped many shares to domestic residents. At the end of 2002, China announced its plan to allow the Qualified Foreign Institutional Investors (QFII) to invest in its "A share" market designed for domestic investors with RMB savings. The Chinese authorities are also studying actively the mechanism of Qualified Domestic Institutional Investors (ODII), which would allow Chinese residents to invest in overseas securities markets, including Hong Kong markets, where many Chinese companies are listed but their shares cannot be sold to Chinese residents through legal channels. Compared to China's inward FDI stock of \$395 billion, its foreign portfolio investment stock of about \$20 billion is insignificant. However, the potential for foreign investment in China's securities market is bright as its stock market capitalisation is almost the same as that of Hong Kong at \$463 billion at the end of 2002, compared to \$2 trillion in Japan and \$11 trillion in the US. China's financial sector is still underdeveloped and is the major bottleneck for China's sustainable growth and development as will be discussed further in the next section. But with China's commitment to WTO and its recent efforts on modernising its financial regulation and development, it is just a matter of time, perhaps as long as another decade, for China's foreign portfolio investment to catch up with its FDI and foreign trade.

To understand the potential foreign portfolio investment in China, it is useful to take a look at the experiences of China's more developed neighbours. This is possible by examining the bilateral long-term securities investment data collected and maintained by the US Treasury. Since the US is the largest player as assets and liabilities holder in cross-border portfolio investment, the picture we get from this data set should be illustrative. Some of the findings, such as China's huge investment in US bonds, are not only surprising as new development but also have important implications for building global capital market order.

Graph 6.1 shows the cross-border gross purchases and sales of stocks and bonds between the US and the rest of the world. Clearly the rest of the world traded much more US stocks and bonds than the US traded the rest of the world's stocks and bonds. In the most recent period, the monthly gross purchases of US bonds by the rest of the world are in the range of \$600 to \$800 billion. The monthly gross purchases of US stocks by the rest of the world are in the range of \$200 to \$400 billion. The monthly gross purchases of foreign bonds or stocks by US residents, on the other hand, have never exceeded \$200 billion. A similar pattern can be seen in Graph 6.3 for the cross-border trading of stocks and bonds between the US and Asia. Asia's love with US bonds went back to the 1980s. Asian countries have accumulated large amounts of net holdings of US bonds, led by Japan and followed now by China.

Table 2.2 to 2.5 show the summary trends on cross-border trading of long-term securities between the US and major Asian economies. The transaction data reported monthly are added up to get the annual numbers. The net purchases are derived by subtracting the gross sales from gross purchases. During the ten years from 1988 to 1997, Asia's net purchases of US bonds reached \$415 billion, compared to \$1.447 billion by the rest of the world. In 2001, Asia's net purchases of US bonds were as high as \$147 billion, compared to \$405 billion by the rest of the world. China's net purchases of US bonds in 2001 were as much as Japan's at about \$52 billion. Both Japan and China have increased their net purchases of US bonds after the Asian financial crisis. During the ten years from 1988 to 1997, China's net purchases of US bonds were only 11.5% of the Asia total. But it increased to 23% in 1999, 19% in 2000, and 35.2% in 2001. Given China's \$280 billion official reserves and about \$260 non-official reserves foreign exchange credit in the banking system, China's increased net purchases of US bonds are inevitable. But it is still surprising to know that China's share is as much as 35.2% of the Asia total.

Table 4.8 also shows a summary account of China's balance of payments since 1982. Two items are related to China's capital outflows. One is the current account surplus and the other is the errors and omissions. China's accumulated current account surplus since 1982 reached \$135 billion while the accumulated errors and omissions since 1982 were even higher at \$140 billion, both at about 12% of GDP. Table 4.8 also shows China's external debt at the end of 2001 was only about \$170 billion or 14.7% of GDP. Clearly China is putting a lot of official and private savings in US dollars. Why? A simple explanation is property rights! Like other foreign investors in US assets, the Chinese government and the Chinese people certainly believe that the property rights of their US investment are well protected. On the other hand, China also provides better protection for property rights in FDI in China than on domestic assets. Hence, on the whole, both sides are happy and better protection of property rights enhances value and productivity of capital.

It is interesting to note that the private foreign bank lending to China is not as important as FDI. This can be seen from the changes in cross-border banking capital flows between Hong Kong and Mainland China during the last decade. As shown in Graph 5.3 and Table 4.11, Hong Kong used to be an important centre in Asia for making syndicated loans to China and other Asian economies. From 1994 to 1999, Hong Kong was a net lender of banking capital to Mainland China. After 2000, however, Hong Kong turned into a net borrower of banking capital from Mainland China. Since 1997, there has been a steady decline in Mainland's gross banking liabilities to Hong Kong from more than \$50 billion in 1997 to less than 20 billion after 2001. This was triggered by the bankruptcy of the GITIC (Guangdong International Trust and Investment Corporation), which borrowed from foreign banks in Hong Kong with the implicit understanding that the Chinese government would guarantee the loans. The Chinese government, however, decided not to use its money to save this regional state-owned holding company in order to avoid moral hazard problem in similar cases for other companies and in the future. After the GITIC bankruptcy, foreign banks became very cautious in extending syndicated loans to China.

During the Asian financial crisis in 1997, Hong Kong suffered a huge withdrawal of foreign banking capital. Hong Kong's foreign banking funds fell from \$630 billion in June 1997 to \$250 billion by April 2002, a drop of 60%. Among the total withdrawal of \$380 billion, \$251 is by Japan (see Table 4.11). In spite of fluctuations in capital flows, Hong Kong's banks have been extremely resilient during and after the crisis with NPLs staying no more than 5%.

During the Asian financial crisis, short-term capital flows were blamed as a driver for financial market instability. Hence, there have been calls for caution on liberalisation of capital accounts and portfolio investment. The role of foreign investors in the stability of local markets becomes an interesting topic. Graphs 6.3 to 6.8 examine this issue using data after the Asian financial crisis and show a strong negative correlation between the size of trading by foreign investors and the level of the local stock market turnover. What it means is that, at least, during the non-crisis period, the participation of foreign investors in the Asian local stock markets enhances the local market stability: the lower the local turnover, the higher the share of trading by the US investors. This pattern appears in Japan, Hong Kong, Singapore, Taiwan, Korea, and even the tiny "B share" market of Mainland China. It appears that the US investment, probably helped by their institutional investors, is more rational than that of the local investors.

# 2 China's Structural and Institutional Conditions for FDI

In spite of the large volume of literature on China's foreign trade and investment by scholars (Lardy, 2002), investment bank economists (Lehman Brothers, 2002 and Goldman Sachs, 2003), and international organisations (OECD, 2002), in our view, the implications of recent development in China relating to its foreign trade and investment are not appreciated properly in their scale, scope and depth. This is partly due to China's many unique development conditions as well as an inclination to use the standard equilibrium tools of economics to deal with intrinsically a disequilibrium problem of development with unlimited supply of cheap labour in China and other developing economies. This section examines this Lewis type dual-sector development issue in order to answer many of the questions raised in the previous questions. The Lewis model of dual-sector development has many new implications when the cross-border mobility of capital and capitalist institutions become possible for a country as large as onefifth of the world in terms of population.

# Development Gaps and Unlimited Supply of Labour

China is large with one-fifth of the world's population. China's labour force is larger than the sum in all developed economies. China's total GDP at purchasing power parity prices is about \$5 trillion, or roughly half of the United States' GDP at current prices. However, China's GDP at current prices is much smaller, about one-fourth of Japan's and one-tenth of US's. However, for some manufacturing products, China's market share can be more than 50% of the global total. The size of China allows it to enjoy economies of scale and scope. China can afford to have all major global auto producers set up joint-venture production bases inside China. However, China's size in monetary terms is limited by the extremely low market prices its labour and products have to face now and for the next decade or so, a fact of life due to its effectively unlimited supply of labour.

As shown in Table 3.4, in global context, China's population is mature, not too young and not too old, with 70% in working age of 15 to 65. China currently has only 7% population above age 65 and 23% under age 15.

China's population compares favourably with both the aging population in the developed economies and the less than mature population in developing economies. The share of working age population in China is among the highest at 70%, compared with only 60% in other less developed countries and 67% in developed countries.

The old age population in the developed countries reaches 15%, or eight percent higher than in China. On the other hand, the population under 15 in other less developed countries is as high as 36%, or 13 percent higher than in China.

Compared with both the developed and the other less developed countries, China's population structure in the past two decades and next two decades is particularly favourable for rapid growth in China, independent of other policy and institutional factors. China's advantage in population structure during these decades is further enhanced by globalisation and the sustained prosperity in the developed economies after World War II. Large amounts of retirement savings from the aging and rich economies accumulated in the last half-century need to be invested in young, growing and productive economies such as China.

China's population structure today is almost the same as Japan's in 1975. Japan's outstanding growth performance during the oneand-half decades since 1975 could suggest that China should have at least one more decade to enjoy the favourable population structure for growth. However, Japan's lost decade since 1990 also reminds China of the challenges ahead. China has a window of opportunities to reform its financial and legal system in the next decade to deal with its future aging problem.

China is poor and has effectively unlimited supply of cheap, mature and educated labour at least for the next one or two decades. China's per capita GDP at purchasing power parity prices of 2000 is \$3,920, slightly higher than the average for other less developed countries at \$3,470, but well below the level of \$22,060 for the developed countries. If using per capita GDP as a rough estimate for average wage, China's average wage at PPP would be about \$1.86 an hour (\$3,920/[12mx22dx8h]). Since China's GDP per capita at current prices is only about \$1,000, China's average wage at current prices would be only \$0.47 an hour, compared to \$10 an hour for the developed economies under the same assumption.

At an average wage of less than half a dollar an hour, China has added 150 million non-farm jobs over the last two decades and reduced the number of people with income less than \$1 a day by 150 million. This number, i.e. 150 million, is as large as the entire population of Western Europe and its Western Offshoots (including US, Canada, and Australia) in 1820 and is more than twice the net migration to the Western Offshoots during the 129 years from 1870 to 1998 (Maddison 2001, p. 28). With this unprecedented achievement in employment creation, China still needs to create at least the same amount (150 million) of non-farm jobs perhaps over the next one to two decades.

China's case clearly fits well into W. Arthur Lewis's dual-sector economic development model of unlimited supply of labour except that the Lewis model has never been applied seriously to an economy as large and as open as the Chinese economy today. Also, China has made great progress not only in providing primary education for most of school age kids but also in higher education. China produces more engineers in one year than Taiwan and Hong Kong could do in ten to twenty years. Hence, China has not only unlimited supply of cheap labour but also unlimited cheap talents. This is confirmed by the fact that it is now more and more difficult for China's college graduates to find good jobs. China's top executive MBA programmes are as expensive as Hong Kong's. China's overseas students have been putting pressures on their classmates from other foreign countries.

## Cross-Border Mobility of Capital and Capitalist Institutions

China's economy is open, in many ways much more open than today's Japanese economy. China attracted much more FDI flows and stocks than Japan. In 2001, Japan, with its half a century long rapid economic growth and development, attracted only \$49 per capita in FDI flows and \$395 per capita in FDI stock, compared to the world average of \$120 in flow and \$1,118 in stock and China's \$37 in flow and \$309 in stock. At official exchange rates, China's foreign trade is more than 40% of GDP while Japan's is about 20%. China allows a large amount of processing trade, which requires a large amount of imported components. Large scale processing trade is only possible for very open economies with close to zero transaction costs, including tariffs and other taxes. China has committed itself to this close to zero transaction costs for processing trade since early 1980s, drawing lessons from its successful neighbours of newly industrialised Asian traders.

China's experiences in opening up its economy have effectively relaxed two crucial assumptions in international trade and development theories: the mobility of capital and the replication of capitalist institutions. Capital is mobile across the Chinese borders on a large scale now as our review in the previous sections show. Traditional trade theories based on factor immobility needs to be modified to take capital mobility into account. For example, capital-intensive industries such as auto and IC manufacturing can be profitable in China because of the capital and technology brought in by FDI and MNCs. The development of these capital-intensive industries is not constrained by the lack of capital. Instead, it is determined by the total costs of internationally linked supply chains and the global demand, including the demand from China itself. Supply chain management theory and global market equilibrium may be more important than the simple application of traditional trade, investment, and macro theories. China's imitation of capitalist economic institutions is also unprecedented in scale, scope, depth, and speed, ranging from central banks, modern public corporations, labour markets, stock markets, and social security systems. The transfer of capitalist institutions and practices is facilitated greatly by the existence of mature market economies in the overseas Chinese communities in Hong Kong and Taiwan as well as large amount of returning overseas students and overseas Chinese business communities. In this regard, the recent success and failure of Japan, Korea and other Asian economies in development and modernisation also set useful examples for China.

It is now possible for MNCs to combine unlimited supply of Chinese cheap labour and cheap talents with advanced international capital, technology and capitalist institutions by relocating some major parts of their production and research in China. MNCs not only benefit from access to China's domestic markets but also gain international competitiveness when they export some of their products from their China bases. Since China allows MNCs to have complete control on their China operations, they can get around China's underdeveloped financial and legal systems, at least for their day-to-day operation. In other words, their risks of operation in China are much lower than in other places in the region.

China's rapid improvement in infrastructure over the last two decades is also an important factor in attracting FDI. This is especially relevant for a few key economic centres along the eastern coast, including the Pearl River Delta region around Guangdong and Hong Kong and the Yangtze River Delta region around Shanghai. China now has more telephone lines than the US. Its highway length is second only to that in the US and will form a complete national network connecting all major cities across China after the current five-year plan. Satellites are used not only for television but also for clearing and settlements in banking and securities transactions. Hong Kong has the world's busiest seaport and airport, handling 60% of China's container shipments. However, major port development planned in Shanghai and Guangzhou are likely to increase their capacity to Hong Kong's current level in one or two decades. This rapid improvement in infrastructure is coordinated by China's forward-looking planning authorities and led by innovative local leaders. China's population structure and its low level of development also allow fast catching up of investment in infrastructure. China's obsession with infrastructure investment is also encouraged by a generation of leaders with engineering backgrounds who see investment in infrastructure as much more valuable than investment in loss-making state-owned enterprises. China's government-directed infrastructure projects may not make profits but will certainly facilitate the mobility of labour, goods, and capital across China's vast western, inland, and coastal regions. They have laid a good physical foundation for China's further integration with the global market economy.

However, in the near future, China's financial and legal systems will come under great pressure to price the risks and returns for millions of large and small projects, which would challenge even the best bankers in the world. In spite of great achievements in legislation, the legal system is still weak in the enforcement of property rights and contracts. This weakness directly affects the robustness and efficiency of China's banking sector and capital markets.

China's financial system is efficient in generating savings but is not effective in allocating them to productive investment. China's banking deposits increased from about 30% to 160% over the last two decades. However, about 20% to 30% of these deposits have difficulty finding productive investment outlets. A substantial part of China's surplus deposits ended up in government bonds. The local governments are then forced to invest these funds in infrastructure projects in order to keep the economy growing at 8% a year.

The flood of savings in China's banking system is partly due to a loose monetary policy during the early 1990s and after the Asian financial crisis. China's money supply (M2) increased ten times from 1990 to 2001 while its nominal GDP rose only five times during the same period. Hence, China's money supply is growing twice as fast as its nominal GDP. China's NPLs is in the range of 25% to 55% of GDP by the official and private estimations. However, China's banking system is still stable for a number of reasons. First, the government is still the owner of most banks in China and is determined to reform its state-owned banks. So, the Chinese people do not feel any risks to their bank deposits. Second, the Chinese economy, although growing at 8% a year, still has ample surplus labour and other unemployed resources, which are looking for employment opportunities following two decades of a gradual but steady loosening of migration policies and control of private enterprises. Third, China's capital account is not open for its domestic RMB deposit holders.

The traditional equilibrium model may not be useful for understanding China's macroeconomic conditions because of China's disequilibrium in the labour markets with unlimited supply of labour. Unlike the situations in Japan and the newly industrialised Asian economies, where the supply of labour quickly hit the limits with wages shooting up, China's market wages for the unskilled labour in major manufacturing centres such as Guangdong have been stagnant at a subsistence level of around \$100 a month for more than a decade.

China's weak and inefficient financial system is creating a puzzle of co-existence of surplus labour and capital. For 16 out of the 21 years since 1982, China ran current account surpluses, including the last 9 years. The accumulated sum since 1982 of China's current account surpluses and the errors and omissions reached 23.8% of GDP in 2001. As China reforms its financial system and opens up its financial sector under the WTO commitment, the efficiency and speed of China's integration into the world economy will be improved steadily. This implies that more and more of China's labour will work directly or indirectly for the global market although their wages will be kept from rising due to ample surplus labour. This trend of development in China is determined by the interaction between China's structural and institutional features and prompts us to further examine the role of China as a world factory and as a regional competitor in the next section.

# 3 Implications for the Rest of the World

To appreciate the impact of China's development and integration into the global economy, it is useful to review the history of global capitalist development as articulated by Angus Maddison. For thousands of years before 1000, the global economy was stagnant but living standards as measured by GDP per capita were pretty much equal across regions. During the first millennium, per capita GDP of the world declined slightly from \$444 to \$435 (at the 1990 international prices for this and all other numbers cited from Maddison, 2001) while world population increased only from 231 to 268 million. During the period from 1000 to 1820, global economic growth picked up but slowly. Although the world population increased from 268 million to 1 billion, average per capita GDP for the world increased only from \$435 to \$667. The world started to diverge in development and income during this period. By 1820, the first world, including Western Europe, Western Offshoots, and Japan, increased their per capita GDP to \$21,470 while the rest of the world, or the third world, only had a per capita GDP level of \$573.

From 1820 to 1998, a great divergence in development and inequality in income emerged between the first and the third world. The share of population in the first world declined from 16.8% to 14.2% but its share of GDP increased from 28.5% to 53.4%. By 1998, world population reached 5.9 billion with 5.1 billion in the third world, including 1.2 billion in China.

The above are the global conditions under which W. Arthur Lewis developed his dual-sector theory on economic development under unlimited supply of labour (Lewis, 1954). Lewis pointed out that the non-capitalist traditional sector in the rural area provides a reservoir for unlimited supply of labour at a fixed urban wage slightly higher than the subsistence rural wage. The urban wage is checked not only by the surplus labour in the rural area but also by rising unemployment in the urban area. This is exactly the condition we see in China now. China's leaders have been working hard to raise the income for peasants. But that is a mission impossible with the Lewis type development reality.

Lewis also points out that the capitalist modern sector is able to generate profits and net savings for investment in capital that is critical for sustained growth. He cites the evidences from Britain where net savings increased from 5% before 1780 to 7% in early 1800s, and 12% around 1870. Similar rises of net savings are also observed in the US between 1840s and 1890s, and in Asia after World War II. Consistent with the Lewis dual-sector model, Britain's Manchester, US's New England, New York, and Chicago, Japan's Tokyo and Osaka, and cities in Asia's newly industrialised economies have become the world factories one by one over time. Now, the centre of global manufacturing is once again on the move towards low costs economies. This time it goes to China's Guangdong and Shanghai, where access to unlimited supply of labour is facilitated by low transportation and declining transaction costs.

#### China as a World Factory: Changes in Global Relative Prices

There is little disagreement on the trend that China will become a world factory. Instead, the debates are on the impact of that trend on the rest of the world. In our view, the single most important impact of China's outstanding development performance is on changes in global relative prices. There is no doubt that new labour supply to the global market will lead to a secular fall in the prices of labourintensive manufacturing products.

Before China's opening up, the world economy was largely a capitalist open market economy dominated by the developed economies. The majority of the third world population, including those in China, were not very relevant to this global market. Let's use the population of age 15 to 65 in Table 3.4 to estimate the impact of China's opening on the labour supply in the world market. The world's working population at age 15 to 65 is 896 million in China, 2,242 million in other less developed countries, and 802 million in more developed countries. Let's assume for simplicity that about onequarter of the labour force in China (224 million) and other less developed countries (560 million) are working for the world market now while the rest are effectively isolated from the world market. Then, the total labour force working for the world market today would be 1,586 million (224m+560m+802m). Since China's labour force was entirely outside of the global market twenty years ago, China's opening has added 224 million to the 1,363 million (560m+802m) labour force that have already been integrated into the global market. That is a net addition of about 16.4% over two decades. If China's past success in integrating its labour into the global market can be continued and replicated in other developing countries, the world market would certainly have no shortage of labour.

It is clear that the additional supply of China's labour force to the global market has concentrated in the manufacturing export sector. This means the global prices for manufacturing products have to fall dramatically. For example, the prices of televisions in the US have been dropping by 8% a year since 1988. The deflation in manufacturing products has started from the traditional labour-intensive products such as toys, plastics, clothing but has been spreading to less labour-intensive products such as electronics and machinery as China advances in its production and technology capability.

As the Say's law predicts, supply creates demand, if there is a perfect market. China's opening leads to growth in income and demand for final products. But it is clearly impossible for China to consume all of its manufacturing products such as televisions, DVDs, motorcycles, and bicycles. Because of China's extremely low level of per capita wealth, the Chinese people's savings rate has been as high as 40.3%, compared to 16.5% in the US, 20.3% in the European Union, 26% in developing countries, and 27.3% in Japan. China's high savings depress its domestic current demand. Hence, export is the most convenient way out and China's export significantly depresses global prices on some specific products such as toys, televisions, and bicycles.

China's impact on deflation in the manufacturing products, however, is not going to cause a global deflation because the weight of manufacturing products made in China is too small in the total expenditure basket of the developed economies. China's exports, which usually incorporate a high proportion of imported components, are only about 4.3% of the world total in 2001. There is a long way to go before China has a direct impact on global deflation.

This is partly due to terms-of-trade effect. The more China exports, the lower the product prices, and the lower the share of China's exports in the total expenditure of the developed economies, other things given.

Anderson and Hu (2003) highlighted that China's net export of manufacturing goods is negligible for causing global deflation in manufacturing prices. This is true but the large deflation effects on specific products are hidden within the net manufacturing exports. The export value is likely to be depressed because of falling prices on China's export while the import value is likely to be high, partly due to stronger demand from China for key components which China can not produce.

In monetary terms, China's impact on the rest of the world is small. But in terms of specific product prices and welfare, or the socalled consumer surplus, China's contribution to the increase of the standards of livings in both developed and developing countries is huge but invisible in statistics. Today's lucky kids around the world can testify to this with their made-in-China toys. Without China's production, most toys may cost five to ten times more.

As we have emphasised, China's impact is on changes in global

relative prices. The deflation in manufacturing sector is accompanied by inflation in commodity and skills, which are needed to support the rapid increase in the volume of manufacturing goods and the rising living standards in China and elsewhere. China has turned from a net exporter of oil to a net importer and its imports of oil have increased steadily in the last decade to about 60 million tons a year by 2001. China's net commodity imports since 1990s have tripled. China's tuition for top rank executive MBA programmes is as high as in Hong Kong or even in the US.

Changes in global relative prices are followed by structural changes in the global economy. The most important change is the shift of manufacturing to China. However, China is only becoming a platform for global manufacturing in order to take advantage of China's favourable labour supply and domestic markets. It is clear that this competition is not between China and the foreign manufacturers. Instead, it is a competition among foreign manufacturers themselves with China only participating as a supply of labour and domestic markets. Foreign invested firms produce half of China's exports with only a small part of the value-added going to China's labour and with a high proportion of imported components. FDI in China concentrated mainly in a few clusters of manufacturing bases in the coastal region (see Table 4.5). The unskilled labour for the export industry is, however, largely coming from poor and remote western and inland regions of China.

China's development and integration with the capitalist world economy is not a miracle if we compare it with the industrialisation in the UK, the US, Japan, and the newly industrialised Asian economies. But the size and structure of China's population and the mobility of capital and capitalist institutions – made possible by today's advances in technology and development knowledge – make China's case special, especially for many of its neighbours.

#### China as a Regional Competitor: Benefits Versus Costs

For many Asians, it is amazing that China weathered through the Asian financial crisis with few changes in its high growth rate of around 8%. Many are wondering whether China's growth is at the costs of others by siphoning off market shares in foreign trade, investment and domestic jobs. There is no doubt that China has become a major competitor in the world market, especially for its

Asian neighbours. The competitive pressure from China can best be seen from Table 1.6 on the share of global inward FDI stock. China increased its share from only 1% in 1980 and 1985 to 5.8% in 2001 while the Asia-7, which has more population than China, did not gain much from 4.2% in 1980 to 4.8% in 2001. China's per capita inward FDI stock reached \$309 by 2001, moving towards Japan's \$395, and well above Asia-7's \$220. However, China's per capita FDI stock is still less than the average for all developing countries, which is at \$478. Table 1.3 shows per capita inward FDI flows and gives similar pattern.

China's competitiveness in labour-intensive manufactures is well recognised and attracted 60% of China's total FDI as shown in Table 4.3. However, FDI is also significant in the non-labour-intensive real-estate sector that has 12% of China's FDI. The services sector also attracted substantial FDI.

What is particularly relevant for our discussion in this section is the concentration of China's FDI in a few clusters of coastal super cities, which have critical mass for global scale production, distribution and financing.

Table 4.5 ranks China's 31 provincial level regions by their FDI inflows in 2001 and provides a number of indicators for the provincial economies. The provinces and cities are then divided into three groups by their ranking in FDI inflows: the top-9, the middle-12, and the bottom-10. The top-9 includes, in descending order of the share of average FDI during 2000-2001, Guangdong (25.7%), Jiangsu (14.9%), Shanghai (9.3%), Fujian (8.5%), Shandong (7.6%), Liaoning (5.4%), Zhejiang (4.8%), Tianjin (4.6%), and Beijing (3.8%). Many foreign visitors are impressed by the physical changes in the cities such as Shanghai and Beijing but the real stars of productive investment and manufacturing capacity in China are Guangdong and Jiangsu, where land prices have not been driven up to international levels as in Hong Kong, Shanghai and Beijing while access to finance, research and other services provided by the big cities is still convenient.

The concentration of economic activities in the top-9 is impressive, if not surprising. This group has about one-third of China's population but produced half of China's GDP, attracted three-quarters of China's FDI, and generated 90% of China's foreign trade.

The middle-12 includes mostly inland provinces while the

bottom-10 consists of all western provinces, the poorest region of China. The middle-12 has half of China's population and one-third of China's GDP but only attracted one-seventh of China's FDI and 8% of China's foreign trade. The bottom-10 has 18% of China's population, 10% of GDP, 3% of foreign trade, and 1.8% of FDI.

Given the diversity of China's regions, it is natural to ask which parts of China we would like to compare with its neighbouring countries. With China's huge population, it is also important to compare FDI per capita.

In 2001, China's top-9 attracted \$97 per capita in FDI, much higher than the Asia-7 of \$12. But the Asia-7 did much better than China's middle-12 at \$10 per capita and bottom-10 at \$4 per capita. Both China's inland and western regions face similar concerns as the Asia-7 countries. They are working hard to improve their investment environment as well as lobby for more support from China's central government. China's central government is also working very hard to help the less developed regions, particularly through the "Go West" strategy. The central government has invested heavily in highways, railroads, airports and other infrastructural works in the western regions but their effects on attracting FDI did not generate the intended results. Instead, the poor people in the western and inland regions continue to rush to the eastern coast for jobs, riding on the new roads built by the government. These unintended results, however, may be better than the plan, as the migrant labourers send remittances, knowledge, and even skills they learnt in the big cities back to their hometowns.

For the Asia-7, labour mobility to China's eastern coast is out of question except for a few highly skilled professionals. But intraindustry trade is expanding as can be seen from the rising trade volume for the six Asian traders (Taiwan, Hong Kong, Korea, Malaysia, Singapore, and Thailand, re-export excluded; see Table 1.2). While China's share of global export increased from 2.5% in 1993 to 4.0% in 2000, the six Asian traders' share increased from 9.7% to 10.5%. During the same period, Asia as a whole increased slightly its share of global exports from 26.3% to 26.7% with a significant decrease in Japan's share from 10% to 7.7% and Australia and New Zealand's share from 1.5% to 1.2%.

There is no doubt that China's export growth is much faster than its neighbours' as shown in Table 4.1. During 1990-2000, China's export growth was 14.9%, compared to 8.4% for the export growth in Asia. But it is due to two factors. First, China started from a very low share of only 2.5% of global trade in 1993. Even in 2001, China's share of global exports was only 4.3%. Second, as discussed in the previous section, China's exports include many components imported from Asian and Western economies. The value-added from China is small.

Table 4.2 compares the market share in global manufactures exports in 1990 and 2000. This table should ease the concerns by China's competitors in Asia a bit as it shows clearly that they can grow with China at the costs of more developed economies in terms of market shares in manufacture export. From 1990 to 2000, the developed economies lost 11 percentage points of market share in global manufactures exports. Asia's gain is as high as 7.3 percentage points. China gained 2.3 percentage points while the six Asian traders gained 3.1 percentage points, more than the gain by China. The rest of Asia excluding China and the six Asian traders also gained 1.3 percentage points.

Hence, China is not only a regional competitor but also an important partner in terms of the integration of Asia's global-scale manufacturing. China not only has an unlimited supply of labour and talents but also large markets for consumer products and manufacturing equipments and parts, which are all opportunities for its neighbours. Hong Kong and Taiwan are the first in seizing the opportunities, followed by the US, Europe, and Japan.

The competitive pressures from China will not go away until most of China's surplus labour is absorbed by the expanding modern sector. The key is how to best position oneself in the increasingly competitive global economy. The pressure to force China to revalue its currency would not affect China's export competitiveness, simply because the real wage in China is not set by the nominal exchange rate. China's wages are extremely flexible at the low end. The real wage for unskilled labour is determined by the subsistence-level income in China's rural areas. The reliable and sustainable way for China's real wage to increase is to help China to develop its poor regions in the western and inland provinces.

However, a revaluation in RMB is likely to put large deflation pressure in the coastal regions' real estate and services sectors as these modern sectors are already substantially integrated with the world economy. The price structure there, except for the wage of unskilled migrant labour, are much more sensitive to changes in nominal exchange rates. In particular, large amount of assets and liabilities concentrated in the more advanced regions of China are denominated in foreign exchange. Any attempt to adjust the currently fixed exchange rate of RMB would be equivalent to redistribution of wealth among holders of foreign exchange assets and liabilities, similar to redistribution among creditors and debtors during inflation or deflation. Given China's sustained surplus in current account and rising official and private foreign exchange reserves, it is entirely creditable for China to maintain the current fixed exchange rate regime. Then, China can leave any adjustments in real wages and other prices to domestic price adjustments. Given China's disequilibria in many markets, including the labour markets, China's fixed exchange rate provides an anchor and a reference for gradually rationalising China's price structure. Partly due to the fixed exchange rate, China's price structure today is largely consistent with the requirement of a market economy.

As discussed in the previous section, China's current account surplus implies that China is exporting capital, which does not seem consistent with China's unlimited supply of labour. Would a revaluation of RMB help to turn China's current account surplus into balance or deficit? Not really! China's excessive savings are not due to exchange rates but are a result of its underdeveloped financial system, which is not able to identify good projects and enforce lending contracts. Revaluation of RMB would depress the best parts of the Chinese economy and lead to less lending and more surplus savings, and hence more current account surplus, just like what happened in Japan when the ven appreciated. So, for those who would like to see a reduction in current account surplus in China or Japan, they need to help China and Japan reform their financial system, and not tamper with the exchange rates. After all, we all have learnt from the Keynesian and Monetarist debates about the neutrality of money and monetary policy in the long run. Exchange rate policy is only part of the monetary policy. The role of China in the region and in the global financial system is real, not just a monetary phenomenon. The aging populations in Japan, Europe and the US also need real returns from their savings, not nominal or monetary illusions.

## 4 Conclusion

This paper reviews global capital flows and the position of China. We have found that the rapid FDI inflows into China, following its economic opening and reform, are essentially driven by two factors: China's unlimited supply of labour and talents and China's declining barriers for cross-border mobility of capital and capitalist institutions.

The combination of China's unlimited supply of labour with foreign capital under capitalist institutions is transforming China into a world factory much like what happened before in Europe, America, and Asia. The consequences of this are also similar to what we have seen in the past: a decline of prices for labour-intensive manufacturing products and a relative rise in the prices for raw materials and skills.

The catching up of China in economic development provides competitive pressures as well as productive opportunities for the world and especially for its neighbours. The aging populations in the developed countries also need to rely on the much younger and mature population in China and Asia to secure good returns on their retirement savings.

However, due to the underdeveloped financial system in China and other developing economies, global savings have refused to flow into these developing economies with growth potential. Even China is having a net export of savings as the Chinese government and people are accumulating foreign assets, especially US bonds and stocks. The global savings and capital flows, although driven by structural factors such as the costs of labour, are hindered by institutional factors, such as the quality of domestic banks and capital markets in the less developed countries. This flight to quality is partly responsible for the tech bubbles in the US and lies at the heart of volatility in global capital flows.

The solution to these global mismatches in capital flows lies not in manipulating exchange rates and other monetary tools, which cannot change the real wages and potential competitiveness of developing economies like China. Instead, developed economies need to focus on real development problems in the developing countries, which have more than 80% of the world's population.

FDI is successful in China largely because foreign investments do not rely on the domestic financial system and foreign investors have complete control over the companies they establish. China has not yet become an engine of growth for the global economy in cash terms. However, in terms of GDP measured by PPP or the welfare of the global population or the global consumer surplus, China is becoming an engine of growth for the world. This is why we need to study China's success and its problems seriously since they have profound implications for all of us.

#### References

- Anderson, Jonathan and Fred Hu (2003), *The Five Myths about China and the World*, Goldman Sachs.
- International Monetary Fund (2002), *World Economic Outlook*, IMF, Washington D.C., September.
- International Monetary Fund (2003), *Global Portfolio Investment Survey*, IMF, Washington D.C.
- Lewis, W. Arthur (1954), *Economic Development with Unlimited* Supply of Labour, The Manchester School.
- Lardy, Nicholas R. (2002), *Integrating China into the Global Economy*, Brookings Institute Press, Washington D.C.
- Maddison, Angus (2001), *The World Economy: A Millennial Perspective*, OECD Development Centre, Paris.
- National Bureau of Statistics (2002), *Statistical Yearbook of China*, China Statistical Press.
- Newton, Alastair and Robert Subbaraman (2002), *China: Gigantic Possibilities, Present Realities*, Lehman Brothers, New York.
- OECD(2002), China in the World Economy: The Domestic Policy Challenges, OECD, Paris.
- Population Reference Bureau (2002), 2002 World Population Data Sheet, Washington D.C.
- US Treasury, "Treasury International Capital System website", http://www.treas.gov/tic/index.html.
- UNCTAD (2002), World Investment Report 2002, UNCTAD, New York/Geneva.
- World Bank (2002), *Global Development Finance*, The World Bank, Washington D.C.
- World Trade Organisation (2001), Annual Report, 2001, WTO, Geneva.
- World Trade Organisation (2002), Annual Report, 2002, WTO, Geneva.

(in billions of dollars and	l percenta	iges)								
	1982	1983	1984	1985	1986	1987	1988	1989	1990	1991
Level										
FDI	59	51	60	58	86	140	164	193	203	160
M&A							116	140	156	81
Service Trade	805	775	800	783	902	1,069	1,218	1,336	1,601	1,673
Merchandise Trade	3,780	3,684	3,889	3,931	4,326	5,050	5,737	6,156	6,982	6,982
Growth										
FDI	-13.2	17.0	-4.3	50.1	61.8	17.1	17.6	17.6	5.3	-21.0
M&A										
Service Trade	-2.4	-3.7	3.2	-2.2	15.3	18.5	13.9	9.7	19.8	4.5
Merchandise Trade	-5.8	-2.5	5.6	1.1	10.1	16.7	13.6	7.3	13.4	0.0
As percentage of Merchandise 7	<b>Frade</b>									
FDI	1.6	1.4	1.5	1.5	2.0	2.8	2.9	3.1	2.9	2.3
M&A							2.0	2.3	2.2	1.2
Service Trade	21.3	21.0	20.6	19.9	20.9	21.2	21.2	21.7	22.9	24.0

Table 1.1. Global Trade and Investment, 1982-2001

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, <u>www.fondad.org</u>

140 Global Capital Flows and the Position of China

	1992	1993	1994	1995	1996	1997	1998	1999	2000	2001	Average 1982- 2001
Level											
FDI	171	228	260	331	386	478	694	1,088	1,492	735	351.9
M&A	79	83	127	187	227	305	532	766	1,144	594	206.2
Service Trade	1,863	1,897	2,077	2,387	2,536	2,627	2,662	2,713	2,872	2,870	1,773.4
Merchandise Trade	7,484	7,464	8,485	10, 142	10,585	10,950	10,797	11,204	12,538	12,601	7,638.3
Growth											
FDI	6.9	32.9	14.1	27.3	16.8	23.8	45.3	56.7	37.1	-50.7	16.3
M&A	-1.8	4.8	53.0	46.8	21.7	34.3	74.4	44.1	49.3	-48.1	27.8
Service Trade	11.4	1.8	9.5	14.9	6.3	3.6	1.3	1.9	5.9	-0.1	6.7
Merchandise Trade	7.2	-0.3	13.7	19.5	4.4	3.5	-1.4	3.8	11.9	0.5	6.1
As percentage of Mer	chandise (	Trade									
FDI	2.3	3.0	3.1	3.3	3.6	4.4	6.4	9.7	11.9	5.8	3.8
M&A	1.2	1.1	1.5	1.8	2.1	2.8	4.9	6.8	9.1	4.7	1.9
Service Trade	24.0	25.4	24.5	23.5	24.0	24.0	24.7	24.2	22.9	22.8	22.8

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, <u>www.fondad.org</u>

Table 1.1. (continued)

Young Rok Cheong and Geng Xiao 141

Source: UNCTAD, World Investment Report 2002, website.

#### Table 1.2 World Merchandise Trade by Region, 1948-2000

	1948	1953	1963	1973	1983	1993	2000
World export in value	58.0	84.0	157.0	579.0	1835.0	3641.0	6186.0
		Expe	ort Share				
World	100.0	100.0	100.0	100.0	100.0	100.0	100.0
North America	27.3	24.2	19.3	16.9	15.4	16.8	17.1
Latin America	12.3	10.5	7.0	4.7	5.8	4.4	5.8
Western Europe	31.5	34.9	41.4	45.4	38.9	43.7	39.5
C./E. Europe/Baltic							
States/CIS	6.0	8.1	11.0	9.1	9.5	2.9	4.4
Africa	7.3	6.5	5.7	4.8	4.4	2.5	2.3
Middle East	2.0	2.7	3.2	4.1	6.8	3.4	4.2
Asia	13.6	13.1	12.4	14.9	19.1	26.3	26.7
Japan	0.4	1.5	3.5	6.4	8.0	10.0	7.7
China	0.9	1.2	1.3	1.0	1.2	2.5	4.0
India	2.2	1.3	1.0	0.5	0.5	0.6	0.7
Australia and							
New Zealand	3.7	3.2	2.4	2.1	1.4	1.5	1.2
Six East Asian trader	s* 3.0	2.7	2.4	3.4	5.8	9.7	10.5
World import in value	e 66.0	84.0	163.0	589.0	1881.0	3752.0	6490.0
		Imp	ort Share				
World	100.0	100.0	100.0	100.0	100.0	100.0	100.0
North America	19.8	19.7	15.5	16.7	17.8	19.8	23.2
Latin America	10.6	9.3	6.8	5.1	4.5	5.2	6.0
Western Europe	40.4	39.4	45.4	47.4	40.0	42.9	39.6
C./E. Europe/Baltic							
States/CIS	5.8	7.6	10.3	8.9	8.4	2.9	3.7
Africa	7.6	7.0	5.5	4.0	4.6	2.6	2.1
Middle East	1.7	2.0	2.3	2.8	6.3	3.2	2.6
Asia	14.2	15.1	14.2	15.1	18.5	23.4	22.8
Japan	1.0	2.9	4.1	6.5	6.7	6.4	5.8
China	1.1	1.7	0.9	0.9	1.1	2.8	3.5
India	3.1	1.4	1.5	0.5	0.7	0.6	0.8
Australia and							
New Zealand	2.6	2.4	2.3	1.6	1.4	1.5	1.3
Six East Asian trader	s* 3.0	3.4	3.1	3.7	6.1	9.9	9.5

(in billions of dollars and percentages)

Note:

\* Asia six: Taiwan, Hong Kong, Korea, Malaysia, Singapore, and Thailand; Significant re-exports excluded. Source: WTO.

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, www.fondad.org

(in billions of e	dollars)													
	1990	1991	1992	1993	1994	1995	1996	1997	1998	1999	2000	2001	Popu- lation (million)	2001 FDI Inflows per capita
Total World	203	160	171	228	260	331	386	478	694	1,088	1,492	735	6,125	120
Developed countries	165	113	107	137	145	203	220	268	484	838	1,227	503	1,197	420
United States	48	23	19	51	45	59	84	103	174	283	301	124	287	433
Japan	2	1	ŝ	0	-	0	0	3	ŝ	13	8	9	127	49
Asia and the Pacific	25	24	33	59	69	76	94	106	96	103	134	102		
China	ŝ	4	11	28	34	36	40	4	4	40	41	47	1,281	37
Hong Kong, China	ŝ	1	4	~	8	9	10	11	15	25	62	23	7	3358
Taiwan Province of China	1	1	1	1	1	2	2	2	0	~	5	4	23	183
Greater China sub-total	8	4	16	35	43	4	53	58	59	68	108	74	1,310	56
India	0	0	0	1	1	2	ŝ	4	ŝ	2	2	3	1,050	3
Indonesia	1	1	2	2	2	4	9	5	0	÷	-5	-3	217	-15
Malaysia	~	4	5	9	5	9	7	9	ŝ	4	4	1	24	23
Philippines	1	1	1	1	2	1	2	1	2	1	1	2	80	22
Republic of Korea	1	1	1	1	1	2	2	3	5	6	6	3	48	99
Singapore	9	5	2	5	6	6	6	11	9	12	5	6	4	2050
Thailand	~	2	2	2	1	2	2	4	5	4	3	4	63	60
Asia-7 sub-total	13	14	13	17	20	26	31	33	24	29	20	18	1,486	12
Developing countries All developing countries	38	44	59	83	109	113	153	191	188	225	238	205	5,018	41
minus China	34	40	48	56	75	77	113	147	144	185	197	158	3,737	42
Source: UNCTAD, World	Intrectme	nt Rehoi	+ 2002	website										

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, <u>www.fondad.org</u>

Table 1.3 FDI Inflows in Selected Regions, 1990-2001

(IIII percentrages)												
	1990	1991	1992	1993	1994	1995	1996	1997	1998	1999	2000	2001
Total World	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0
Developed countries	81.2	70.6	62.7	60.3	55.7	61.5	57.0	56.0	69.7	77.0	82.3	68.4
United States	23.9	14.2	11.2	22.3	17.4	17.8	21.9	21.6	25.1	26.0	20.2	16.9
Japan	0.0	0.9	1.6	0.0	0.3	0.0	0.1	0.7	0.5	1.2	0.6	0.8
Asia and the Pacific	12.2	15.2	19.4	26.0	26.5	23.0	24.3	22.2	13.9	9.5	9.0	13.9
China	1.7	2.7	6.5	12.1	13.0	10.8	10.4	9.3	6.3	3.7	2.7	6.4
Hong Kong, China	1.6	0.6	2.3	3.0	3.0	1.9	2.7	2.4	2.1	2.3	4.2	3.1
Taiwan Province of China	0.7	0.8	0.5	0.4	0.5	0.5	0.5	0.5	0.0	0.3	0.3	0.6
Greater China sub-total	4.0	4.2	9.3	15.5	16.6	13.2	13.6	12.1	8.5	6.2	7.2	10.0
Singapore	2.7	3.1	1.3	2.1	3.3	2.7	2.2	2.2	0.9	1.1	0.4	1.2
Republic of Korea	0.4	0.7	0.4	0.3	0.3	0.5	0.6	0.6	0.8	0.9	0.6	0.4
India	0.1	0.0	0.1	0.2	0.4	0.7	0.7	0.8	0.4	0.2	0.2	0.5
Indonesia	0.5	0.9	1.0	0.9	0.8	1.3	1.6	1.0	-0.1	-0.3	-0.3	-0.4
Malaysia	1.3	2.5	3.0	2.5	1.8	1.8	1.9	1.3	0.4	0.4	0.3	0.1
Philippines	0.3	0.3	0.5	0.5	0.6	0.4	0.4	0.3	0.3	0.1	0.1	0.2
Thailand	1.3	1.3	1.2	0.8	0.5	0.6	0.6	0.8	0.7	0.3	0.2	0.5
Asia-7 sub-total	6.6	8.9	7.6	7.3	7.7	8.0	8.0	6.9	3.4	2.6	1.4	2.5
Developing countries	18.5	27.7	34.6	36.6	41.9	34.0	39.5	40.0	27.0	20.7	15.9	27.9
All developing countries minus China	16.8	25.0	28.1	24.5	28.8	23.2	29.1	30.7	20.7	17.0	13.2	21.5

Table 1.4 Share of Global FDI Inflows by Selected Regions, 1990-2001

(in percen

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, www.fondad.org

Source: UNCTAD, World Investment Report 2002, website.

(in percentages)											
	1991	1992	1993	1994	1995	1996	1997	1998	1999	2000	2001
Total World	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0
Developed countries	91.7	86.2	81.4	87.0	87.9	82.6	76.1	83.4	88.7	92.3	83.5
United States	35.0	20.0	24.1	35.2	28.5	30.0	26.8	39.4	32.9	28.4	31.1
Japan	0.2	0.3	0.1	0.6	0.3	0.8	1.0	0.8	2.1	1.4	2.6
Asia and the Pacific	2.7	4.6	8.8	3.7	3.8	5.9	7.1	3.0	3.8	1.9	5.8
China	0.2	0.3	0.7	0.6	0.2	0.8	9.0	0.2	0.3	0.2	0.4
Hong Kong, China	0.7	2.1	6.4	1.3	0.9	1.4	2.4	0.2	0.5	0.4	1.7
Taiwan Province of China							0.2	0.0	0.2	0.1	0.4
Greater China sub-total							3.2	0.3	1.1	0.7	2.6
Singapore			0.1	0.3	0.1	0.1	0.5	0.1	0.1	0.1	0.2
Republic of Korea	0.2	0.3	0.2	0.2	0.4	0.2	0.1	0.1	0.2	0.1	0.6
India	0.2	0.1	0.6	0.3	0.1	0.3	0.1	0.2	0.2	0.0	0.2
Indonesia	0.1	0.5	0.2	0.7	0.6	0.2	1.4	0.4	0.2	0.0	0.3
Malaysia	0.8	0.0	0.0	0.0	0.1	0.2	0.3	0.7	1.3	0.6	0.6
Philippines	0.3	0.3	0.4	0.3	0.7	0.3	0.1	0.1	0.4	0.1	0.8
Thailand	0.1	0.6	0.1	0.1	0.1	0.1	0.2	0.6	0.3	0.2	0.2
Asia-7 sub-total			1.6	1.8	2.1	1.5	2.7	2.2	2.6	1.2	3.0
Developing countries	7.2	10.3	17.2	11.8	8.8	15.7	22.0	15.5	9.7	6.2	14.4

Source: UNCTAD, World Investment Report 2002, website.

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, <u>www.fondad.org</u>

Table 1.5 Shares in M&A Sales by Selected Regions and Countries

1980-2001	
Countries,	
<b>Regions and</b>	
by Selected	-
nward Stock	11. 2 1 11
.6 FDI 1	•
Table1.	

2	ges)
	Itag
	cen
	Jer.
	l pr
•	ar
	llars
	qo
	of
	ions
	mill
	н.

	1980	1985	1990	1995	2000	2001	1980	1985	1990	1995	2000	2001	Popu- lation (million) P	Inward FDI Stock er capita
Total World Developed countries United States Japan	635,534 389,715 83,046 3,270	913,182 568,670 184,615 4,740	$1,871,594 \\1,382,978 \\394,911 \\9,850$	2,911,725 2,021,303 535,553 33,508	6,258,263 4,124,261 1,214,254 50,323	6,845,723 4,504,122 1,321,063 50,319	$ \begin{array}{c} 100.0 \\ 61.3 \\ 13.1 \\ 0.5 \end{array} $	100.0 62.3 20.2 0.5	$   \begin{array}{c}     100.0 \\     73.9 \\     21.1 \\     0.5   \end{array} $	$   \begin{array}{r}     100.0 \\     69.4 \\     18.4 \\     1.2   \end{array} $	$   \begin{array}{r}     100.0 \\     65.9 \\     19.4 \\     0.8   \end{array} $	100.0 65.8 19.3 0.7	6,125 1,197 287 127	$ \begin{array}{c} 1,118\\ 3,763\\ 4,597\\ 395 \end{array} $
Asia and the Pacific China Hong Kong, China Taiwan Province of China	161,1966,251124,2862,405	$\begin{array}{c} 228,970\\ 10,499\\ 129,750\\ 2,930\end{array}$	317,663 24,762 148,183 9,735	570,625 137,435 174,063 15,736	$1,246,700 \\ 348,346 \\ 429,036 \\ 27,924$	1,329,431 $395,192$ $451,870$ $32,033$	25.4 1.0 19.6 0.4	25.1 1.1 14.2 0.3	$17.0 \\ 1.3 \\ 7.9 \\ 0.5$	19.6 4.7 6.0 0.5	19.9 5.6 6.9 0.4	19.4 5.8 6.6 0.5	1,281 7 23	309 66,451 1,424
Greater China sub-total Singapore Republic of Korea India Indonesia Malaysia Philippines Thailand Asia-7 sub-total	132,942 1,177 10,274 5,169 1,281 1,327 6,203 981 981 26,412	$\begin{array}{c} 143,179\\ 1,075\\ 2,971\\ 7,388\\ 2,601\\ 2,160\\ 13,016\\ 1,999\\ 53,210\\ 53,210\\ \end{array}$	182,680 1,668 38,883 38,883 10,318 3,268 5,864 2,864 8,209 8,209 96,774	327,234 5,652 50,601 28,732 6,086 9,991 59,582 17,452 178,095	805,306 18,916 60,638 52,748 12,440 62,786 62,786 95,714 24,468 327,709	879,095 22,319 57,361 53,302 14,232 47,228 104,323 28,227 326,991	20.9 0.2 0.2 0.3 0.2 0.2 0.2 0.2 0.2 0.2 0.2	$\begin{array}{c} 15.7 \\ 0.1 \\ 2.7 \\ 0.8 \\ 0.3 \\ 0.2 \\ 0.2 \\ 0.2 \\ 5.8 \end{array}$	9.8 0.1 0.6 0.2 0.3 1.5 7.2	$\begin{array}{c} 11.2 \\ 0.2 \\ 1.7 \\ 1.0 \\ 0.2 \\ 0.3 \\ 0.6 \\ 0.6 \\ 6.1 \end{array}$	$\begin{array}{c} 12.9\\ 0.3\\ 0.8\\ 0.2\\ 0.2\\ 1.0\\ 1.5\\ 5.2\end{array}$	$\begin{array}{c} 12.8\\ 0.3\\ 0.8\\ 0.8\\ 0.2\\ 0.7\\ 1.5\\ 1.5\\ 4.8\\ 4.8\end{array}$	$\begin{array}{c} 1,310\\ 1,050\\ 217\\ 217\\ 24\\ 80\\ 48\\ 63\\ 1,486\\ 1,486\end{array}$	671 21 264 2,185 178 178 976 24,839 451 451 220
Developing countries All developing countries minus China	245,819 239,568	344,463 333,964	484,954 460,193	849,915 712,480	2,002,173 1,653,827	2,181,249 1,786,057	38.7 37.7	37.7 36.6	25.9 24.6	29.2 24.5	32.0 26.4	31.9 26.1	5,018 3,737	435 478

Source: UNCTAD, World Investment Report 2002, website.

146 Global Capital Flows and the Position of China

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, <u>www.fondad.org</u>

			( <b>D</b>											
	1980	1985	1990	1995	2000	2001	1980	1985	1990	1995	2000	2001	Popu- lation (million) P	Inward FDI Stock er capita
Total World Developed countries	521,486 499,428 215 275	691,745 656,276 238 360	1,721,462 1,630,443 430,521	2,854,853 2,577,550 600.015	6,086,428 5,316,292 1 203 431	6,552,011 5,751,947 1 381 674	100.0 95.8 41.2	100.0 94.9 34.5	100.0 94.7	100.0 90.3	100.0 87.3 21.2	100.0 87.8 71.1	6,125 1,197 287	1,070 4,805 4 807
Japan	19,610	43,970	201,440	238,452	278,445	300,115	3.8	6.4 6.4	11.7	6.4.2 4.8	4.6	4.6	127	т, <sup>оо</sup> / 2,356
Asia and the Pacific	6,206	11,699	47,813	185,931	583,524	603,290	1.2	1.7	2.8	6.5	9.6	9.2		
China $U_{11} = V_{22} = O_{122}$	- 140	131	2,489	15,802	25,804	27,579	0	0.0	0.1	0.6 2.0	0.4	0.4 1	1,281	22
Taiwan Province of China	041 04	204 204	11,920 12,888	25,144	49.187	54.667	0.0	0.0	0.7	0.7 0.9	0.0	0.8	23	2.430
Greater China sub-total	2,679	27,297	119,779	440,794	457,026	0.0	0.4	1.6	4.2	7.2	7.0	1,310	349	<b>.</b>
Singapore	235	250	281	495	1,311	2,068	0.0	0.0	0.0	0.0	0.0	0.0	1,050	2
Republic of Korea	'	55	27	1,295	2,339	2,464		0.0	0.0	0.0	0.0	0.0	217	11
India	197	1,374	2,671	11,143	18,688	18,955	0.0	0.2	0.2	0.4	0.3	0.3	24	777
Indonesia	171	171	155	1,220	1,965	2,126	0.0	0.0	0.0	0.0	0.0	0.0	80	27
Malaysia	127	461	2,301	7,787	50,552	40,825	0.0	0.1	0.1	0.3	0.8	0.6	48	844
Philippines	3,718	4,387	7,808	35,050	53,009	63,225	0.7	0.6	0.5	1.2	0.9	1.0	4	15,054
Thailand	13	14	404	2,173	2,439	2,610	0.0	0.0	0.0	0.1	0.0	0.0	63	42
Asia-7 sub-total	6,712	13,698	59,163	130,304	132,274	0.9	1.0	0.8	2.1	2.1	2.0	1,486	89	
Developing countries All developing countries	22,058	35,469	90,404	270,925	751,632	776,065	4.2	5.1	5.3	9.5	12.3	11.8	5,018	155
minus China	22,058	35,338	87,915	255,123	725,828	748,486	4.2	5.1	5.1	8.9	11.9	11.4	3,737	200
Source: UNCTAD, Wo	rld Investm	tent Repo	<i>rt 2002</i> , w	ebsite.										

 Table 1.7 FDI Outward Stock in Selected Regions and Countries, 1980-2001

 (in millions of dollars and necentages)

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, <u>www.fondad.org</u>

Young Rok Cheong and Geng Xiao 147

iabilities in Selected Countries	
Investment I	2001
Derived Portfolio ]	Year-End 1997 and
Table 2.1	

(in millions of dollars and percentages)

	Equity 5	Securities	Long-term <b>E</b>	Jebt Securities	Short-term	Debt Securities	TC	TAL	Share in <sup>7</sup>	FOTAL
Investment in:	1997	2001	1997	2001						
United States	427,579	997,821	886,325	1,653,419	36,192	417,850	1,350,096	3,069,090	22.18	24.46
United Kingdom	313,962	705,331	232,378	395,148	7,099	153,795	553,439	1,254,346	9.09	10.00
Germany	143,058	271,367	446,255	798,524	2,437	81,488	591,750	1,151,378	9.72	9.18
France	163,195	388,422	105, 150	332,358	2,222	55,209	270,567	775,990	4.44	6.19
Netherlands	164,443	285,897	116,817	368,280	1,938	37,417	283,199	691,594	4.65	5.51
Japan	241,804	333,581	144,855	157,246	7,382	36,553	394,041	527,380	6.47	4.20
Hong Kong SAR of China	62,952	79,056	9,884	8,829	1,188	505	74,024	88,390	1.22	0.70
Korea, Republic of	6,085	51,666	31,259	22,650	706	2,018	38,050	76,334	0.62	0.61
Singapore	19,316	35,850	4,918	7,330	129	1,236	24,363	44,417	0.40	0.35
Taiwan Province of China	9,302	38,808	2,545	2,004	21	65	11,868	40,877	0.19	0.33
Russian Federation	10,937	10,753	17,530	14,831	1,687	318	30,154	25,902	0.50	0.21
Malaysia	14,645	11,508	10,541	9,526	417	293	25,603	21,327	0.42	0.17
China, P.R.	4,618	13,210	14,112	5,504	610	1,457	19,340	20,103	0.32	0.16
India	10,396	13,252	4,404	1,793	203	214	15,003	15,260	0.25	0.12
Philippines	4,658	3,452	7,206	8,823	98	332	11,962	12,607	0.20	0.10
Thailand	4,526	7,684	8,108	3,837	276	349	12,909	11,870	0.21	0.09
Indonesia	4,258	3,800	5,226	1,607	870	64	10,354	5,486	0.17	0.04
Total value of investment	2,567,784	5,134,498	3,421,999	6,373,367	98,430	1,038,297	6,088,217	12,546,226	100.00	100.00

Source: IMF, Global Portforlio Investment Survey, 2003.

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, www.fondad.org

	All-ex-US	Asia	Iapan	Hong	Singa-	Taiwan	Korea	China
			5.1	Kong	pore			
			Ŀ	1mount				
1988-1997	349,729	127,033	80,211	19,773	4,625	202	8,737	1,400
1998	-6,212	8,594	3,694	1,385	929	487	1,907	8
1999	-15,640	46,873	46,134	-2,777	-149	1,767	1,965	222
2000	13,088	-11,198	-16,461	3,254	-3,038	767	2,057	251
2001	50,113	27,523	19,938	4,823	-2,487	2,949	2,006	-40
			As perceng	ate of Asia T	<i>fotal</i>			
1988-1997	275.3	100.0	63.1	15.6	3.6	0.2	6.9	1.1
1998	-72.3	100.0	43.0	16.1	10.8	5.7	22.2	0.1
1999	-33.4	100.0	98.4	-5.9	-0.3	3.8	4.2	0.5
2000	116.9	-100.0	-147.0	29.1	-27.1	6.8	18.4	2.2
2001	182.1	100.0	72.4	17.5	-9.0	10.7	7.3	-0.1

 Table 2.2 Net Purchases of Foreign Stocks by US Residents

(in millions of dollars and percentages)

Source: US Treasury website.

Table 2.3 Net Purchases of Foreign Bonds by US Residen	ts
--	----

(in millions of dollars and percentages)

	All-ex-US	Asia	Japan	Hong	Singa-	Taiwan	Korea	China
				Kong	pore			
			1	4mount				
1988-1997	302,824	15,001	-865	-10,045	-2,532	-7,619	9,627	-825
1998	17,349	-4,602	-1,952	-2,452	-2,445	-815	3,161	-1,716
1999	5,676	-3,912	-2,497	-1,458	-334	-2,173	-719	-336
2000	4,054	-13,290	-4,509	-984	-893	-2,762	-1,365	-1,808
2001	-30,393	-15,654	178	-3,298	-293	-3,792	-1,856	-4,033
			As perceng	gate of Asia T	<i>fotal</i>			
1988-1997	2018.7	100.0	-5.8	-67.0	-16.9	-50.8	64.2	-5.5
1998	377.0	-100.0	-42.4	-53.3	-53.1	-17.7	68.7	-37.3
1999	145.1	-100.0	-63.8	-37.3	-8.5	-55.5	-18.4	-8.6
2000	30.5	-100.0	-33.9	-7.4	-6.7	-20.8	-10.3	-13.6
2001	-194.2	-100.0	1.1	-21.1	-1.9	-24.2	-11.9	-25.8

Source: US Treasury website.

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, <u>www.fondad.org</u>

	All-ex-US	Asia	Japan	Hong Kong	Singa- pore	Taiwan	Korea	China
			A	mount				
1988-1997	115,571	9,615	6,617	1,031	8,828	409	-29	28
1998	50,020	-13,781	-1,171	-2,223	-8,438	-69	-84	1
1999	107,522	3,379	5,723	-156	-852	37	-78	204
2000	174,890	21,683	2,070	215	10,788	-147	-160	-103
2001	116,386	22,516	6,788	675	13,078	261	-76	3
			As perceng	ate of Asia T	Fot <i>al</i>			
1988-1997	1202.0	100.0	68.8	10.7	91.8	4.3	-0.3	0.3
1998	363.0	-100.0	-8.5	-16.1	-61.2	-0.5	-0.6	0.0
1999	3182.1	100.0	169.4	-4.6	-25.2	1.1	-2.3	6.0
2000	806.6	100.0	9.5	1.0	49.8	-0.7	-0.7	-0.5
2001	516.9	100.0	30.1	3.0	58.1	1.2	-0.3	0.0

# Table 2.4 Net Purchases of US Stocks by Foreign Residents

(in millions of dollars and percentages)

Source: US Treasury website.

Table 2.5	Net Purchases of US Bonds by Foreign Residents
	(in millions of dollars and percentages)

	All-ex-US	Asia	Japan	Hong Kong	Singa- pore	Taiwan	Korea	China
			A	1mount				
1988-1997	1,447,448	415,223	202,793	52,694	36,900	22,701	2,900	47,890
1998	227,771	45,092	21,432	9,223	9,935	-2,996	15,812	3,519
1999	242,639	74,155	37,643	6,844	-7,417	-483	11,273	17,053
2000	282,938	82,474	49,936	10,181	-4,574	-5,240	5,839	15,656
2001	405,413	147,141	51,873	29,274	389	9,930	533	51,784
			As perceng	ate of Asia T	<i>fotal</i>			
1988-1997	348.6	100.0	48.8	12.7	8.9	5.5	0.7	11.5
1998	505.1	100.0	47.5	20.5	22.0	-6.6	35.1	7.8
1999	327.2	100.0	50.8	9.2	-10.0	-0.7	15.2	23.0
2000	343.1	100.0	60.5	12.3	-5.5	-6.4	7.1	19.0
2001	275.5	100.0	35.3	19.9	0.3	6.7	0.4	35.2

(in millions of dollars and percentages)

Source: US Treasury website.

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, <u>www.fondad.org</u>

	Regular	Preferential rate for foreign investment
Korea	27	
Singapore	22	
Hong Kong	16	
China	33	15
USA	35	
Germany	25	
Japan	30	

#### Table 3.1 Corporate tax comparison among Asian countries (in percentages)

Source: Joongang Daily, March 5, 2003, p. 3.

	1990	1994	1995	1996	1997	1998	1999	2000	Average
Singapore	73.5	79.3	81.8	83.2	84.5	89.2	89.5	86.8	83.5
Hong Kong	33.8	37.7	39.8	41.4	54.3	55.1	60.9	66.2	48.7
Taiwan	44.5	37.2	33.6	31.0	28.5	33.6	36.6	34.0	34.9
Malaysia	21.8	32.6	26.2	26.1	27.6	33.0	37.6	31.8	29.6
Yemen	12.7	17.3	17.8	17.2	16.3	18.6			16.7
Venezuela	17.1	14.5	12.1	18.0	16.3	12.5	12.1	10.7	14.2
Norway	12.4	14.0	14.4	16.0	14.7	11.6	12.9	12.2	13.5
Swiss	12.8	12.3	11.0	13.5	14.5	13.9	14.1	12.5	13.1
China	7.4	9.3	10.5	12.8	15.5	15.0	15.6	15.3	12.7
Indonesia	6.9	6.8	6.8	8.0	11.9	18.8	16.8	21.0	12.1
Korea	5.7	6.1	6.6	6.7	7.4	14.1	17.4	23.4	10.9
Saudi Arabia	8.2	4.9	5.6	9.1	9.2	9.9	10.8	10.4	8.5
Spain	10.0	8.2	5.7	9.9	12.9	9.1	5.5	5.2	8.3
Austria	5.5	7.8	7.6	9.8	9.4	9.4	7.1	7.0	8.0
Denmark	7.8	5.3	5.6	7.5	11.1	7.5	12.7	0.9	7.3
group average	7.8	7.1	6.3	9.1	11.1	8.7	8.4	4.4	7.9
Sweden	7.6	10.5	8.9	7.1	4.2	5.3	5.8	6.3	7.0
Finland	6.8	9.1	7.2	4.9	6.4	6.3	5.6	6.0	6.5
Brazil	1.7	9.0	7.5	7.8	6.5	5.6	6.5	5.8	6.3
Mexico	3.8	2.3	6.3	6.0	7.2	8.1	6.4	6.2	5.8
India	0.4	6.0	5.2	5.2	6.3	6.5	7.1		5.3
Netherlands	5.7	8.9	7.5	6.1	6.0	4.3	1.7	1.9	5.2
Belgium	5.7	5.2	5.3	5.9	6.1	6.0	3.5	3.5	5.2
France	5.3	5.2	4.9	5.0	5.1	4.2	3.9	3.8	4.7
Japan	2.3	2.3	3.6	4.7	5.2	4.6	5.5	7.8	4.5
Australia	5.3	3.0	3.1	3.4	4.4	3.8	4.9	4.6	4.1
Italy	5.5	3.0	2.9	3.5	4.7	2.0	1.7	2.1	3.2
Canada	2.8	1.9	2.1	3.0	2.5	3.4	3.6	4.1	2.9
Great Britain	3.3	3.6	3.5	2.9	2.2	1.9	2.1	2.8	2.8
Germany	2.3	1.1	0.9	1.0	1.4	1.7	1.7	1.7	1.5
USA	0.9	0.6	0.7	0.5	0.4	0.0	0.3	0.3	0.5
Average	11.3	12.2	11.8	12.6	13.4	13.8	14.1	14.1	12.9

Table 3.2 Foreign Exchange Reserve to GDP Ratio

	1990 Rank	1999 Rank	1990	1999	2000	2001
USA	1	1	5,743.8	9,268.6	9,872.9	10,082.2
Iapan	2	2	2,970.1	5.015.0	4,454.6	4.172.5
Germany	3	3	1.503.6	1.991.5	1.891.0	1.845.3
United Ki	ngdom 6	4	983.6	1,448.3	1,397.6	1,423.7
France	4	5	1,195.4	1,355.7	1,308.9	1311
Italy	5	6	1,094.0	1,112.9	1,084.7	
China	10	7	387.8	991.1	1,080.1	1,139.8
Canada	7	8	573.8	675.7	703.9	705.6
Spain	8	9	492.0	565.6	563.9	
Brazil	9	10	442.9	538.8	555.9	
Mexico	15	11	247.0	482.3	567.5	
India	11	12	306.0	450.0	0.0	
Korea	14	13	253.7	424.2	408.9	422.2
Australia	12	14	294.8	397.7	361.1	357.1
Netherlan	ids 13	15	283.5	375.3	373.2	
Taiwan	19	16	162.7	290.5	313.9	282.4
Switzerlan	nd 17	17	228.4	242.9	247.1	
Belgium	18	18	193.8	237.1	230.8	
Sweden	16	19	229.8	231.3	218.4	
Austria	20	20	158.4	198.0	191.7	
Denmark	22	21	129.1	166.2	163.6	
Hong Kor	ng 26	22	72.6	158.0	162.5	
Indonesia	24	23	106.1	156.7	134.5	
Norway	23	24	115.5	148.4	158.6	
Poland	27	25	59.0	148.3	165.5	
Saudi Aral	bia 25	26	104.7	142.9	173.3	
Finland	21	27	134.8	121.0	122.5	
Venezuala	28	28	48.6	96.5	117.8	
Singapore	31	29	37.5	85.3	91.9	85.6
Malaysia	29	30	42.8	78.9	89.9	
Nigeria	33	31	32.4	34.5	36.5	
Kuwait	34	32	18.2	29.8	38.0	
Romania	30	33	38.3	29.6	30.7	
Yemen	32	34	33.6	0.0	0.0	

 Table 3.3 GDP by Country (at current price and official exchange rate)

		Hat	itants			Share in Regio	n Total
	All	Below 15	15 to 65	Above 65	Below 15	15 to 65	Above 65
China	1,281	295	896	06	23	70	7
Less developed countries excluding China	3,737	1,345	2,242	149	36	60	4
More developed countries	1,197	215	802	180	18	67	15
East Asia	1,512	333	1,058	121	22	70	8
U.S.	287	60	190	37	21	99	13
Western Europe	184	31	123	29	17	67	16
Japan	127	18	87	23	14	68	18

8
5
Sheet,
Data
Population
1
$W_{orb}$
e Bureau,
Reference Bureau,
Population Reference Bureau,

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, www.fondad.org
	Merchandi	se Exports	Merchandi	se Imports
	World Total in 2000	Growth 1990-2000	World Total in 2000	Growth 1990-2000
World	100.0	6.0	100.0	6.0
Asia	26.7	8.4	22.8	7.6
Japan	7.7	5.2	5.9	4.9
China	4.0	14.9	3.5	15.5
Hong Kong	3.3	n.a.	3.3	n.a.
	Commerci Exp	al Service orts	Commerci Imp	al Service orts
	World Total in 2000	Growth 1990-2000	World Total in 2000	Growth 1990-2000
World	100.0	6.0	100.0	6.0
Asia	21.1	9.0	25.4	7.0
Japan	4.7	5.0	8.1	3.0
China	2.1	18.0	2.5	24.0
Hong Kong	2.9	0.09	1.8	0.09

#### Table 4.1 China in World Trade

(in percentages)

Source: WTO.

## Table 4.2 Share in World Exports of Manufactures, 1990 and 2000 (in percentages)

	1990	2000	Gain/Loss
World	100.0	100.0	
Developed countries	80.4	69.4	-11.0
Developing countries	17.5	27.4	9.9
Asia	12.6	19.9	7.3
China	1.9	4.7	2.8
Asia-Six*	9.1	12.2	3.1
Other Asia	1.6	2.9	1.3

Note:

\* Asia Six: Taiwan, Hong Kong, Korea, Malaysia, Singapore and Thailand; Significant re-exports excluded. *Source:* WTO.

	-					
		Value			Share	
Sector	1999	2000	2001	1999	2000	2001
Manufacturing	22,603	25,844	30,907	56.1	63.5	65.9
Electric Power, Gas and Water Production and Supply	3,703	2,242	2,273	9.2	5.5	4.8
Wholesale & Retail Trade and Catering Services	965	858	1,169	2.4	2.1	2.5
Transport, Storage, Post and Telecommunication services	1,551	1,012	909	3.8	2.5	1.9
Farming, Forestry, Animal Husbandry and Fishery	710	676	899	1.8	1.7	1.9
Mining and Quarrying	557	583	811	1.4	1.4	1.7
Construction	917	905	807	2.3	2.2	1.7
Scientific Research and Polytechnical Services	110	57	120	0.3	0.1	0.3
Health Care, Sports and Social Welfare	148	106	119	0.4	0.3	0.3
Education, Culture and Arts, Radio, Film and Television	61	54	36	0.2	0.1	0.1
Banking and Insurance	98	76	35	0.2	0.2	0.1
Geological Prospecting and						
Water Conservancy	5	5	10	0.0	0.0	0.0
Other Sectors	753	1,453	1,051	1.9	3.6	2.2

#### Table 4.3 Foreign Direct Investment in China by Sector

(in millions of dollars and percentages)

	1995	1996	1997	1998	1999	2000	2001
				Level	!		
Hong Kong and Macao	20.5	21.3	21.0	18.9	16.7	15.9	17.0
Japan	3.1	3.7	4.3	3.4	3.0	2.9	4.4
Taiwan	3.2	3.5	3.3	2.9	2.6	2.3	3.0
Singapore	1.9	2.2	2.6	3.4	2.6	2.2	2.1
Korea	1.0	1.4	2.1	1.8	1.3	1.5	2.2
Neighboring Countries	29.7	32.0	33.4	30.5	26.2	24.7	28.7
USĂ	3.1	3.4	3.2	3.9	4.2	4.4	4.4
Virgin Island	0.3	0.5	1.7	4.0	2.7	3.8	5.0
Great Britain	0.9	1.3	1.9	1.2	1.0	1.2	1.1
Germany	0.4	0.5	1.0	0.7	1.4	1.0	1.2
Others	3.2	3.9	4.1	5.2	4.9	5.6	6.5
Total	37.5	41.7	45.3	45.5	40.3	40.7	46.9
			S	Share of	total		
Hong Kong and Macao	54.7	51.0	46.5	41.6	41.4	38.9	36.3
Iapan	8.3	8.8	9.6	7.5	7.4	7.2	9.3
Taiwan	8.4	8.3	7.3	6.4	6.5	5.6	6.4
Singapore	4.9	5.4	5.8	7.5	6.5	5.3	4.6
Korea	2.8	3.3	4.7	4.0	3.2	3.7	4.6
Neighboring Countries	79.1	76.7	73.8	67.0	64.9	60.7	61.1
USĂ	8.2	8.2	7.2	8.6	10.5	10.8	9.4
Virgin Island	0.8	1.3	3.8	8.9	6.6	9.4	10.7
Great Britain	2.4	3.1	4.1	2.6	2.6	2.8	2.2
Germany	1.0	1.2	2.2	1.6	3.4	2.6	2.6
Others	8.5	9.4	9.0	11.4	12.1	13.7	13.9
Total	100.0	100.0	100.0	100.0	100.0	100.0	100.0

#### Table 4.4 China's FDI Inflows by Source Country, 1995-2001

(in billions of dollars and percentages)

Source: National Bureau of Statistics, Statistical Yearbook of China, 2002.

	GDP per capita (2001, current price)	1,008	1,648 1,558 3,696	1,490	1,446 1,762 2,208	2,479 1,647	940 728 1 003	617 826	626 561	1,126	626 756	681 655	784 502	693 777	954 954	636 636	
	FDI per capita (2001)	36	153 94 266	114 39	60 212 212	$^{128}_{97}$	20 112 10	20	~~~	10 13 9 13	10	81	vwc	11		0 1 4	
	FDI as share of Fixed Capital Formation (2001)	23.7	75.0 49.6 49.7	80.9	32.0 51.3 51.3	26.8 43.3	14.3 15.9 6.6	35.1	14.5 11.6	6.1 4.3 8.5	7.0 11.4	8.3 5.9	3.6 2.1	2.6	0.4	0.0	
	Trade contribution by FIEs (2000-2001 average)	50.4	53.6 62.1 60.8	61.4 49.8	59.6 31.0 79.2	31.7 54.0	29.3 17.5 79.9	21.0 45.8	15.5 15.5 23.2	14.3 11.4	26.4 24.4	16.1 11.5	9.0 10.1	10.1 9.9	0.4	3.2	
	Trade share (2000-2001 average)	100.0	$36.1 \\ 10.5 \\ 11.7 \\ $	4.8 6.2	4.2 3.6 3.6	5.3 89.4	0.8 0.6 1 1	0.0	.00 4.0	0.5	2.7	0.6 0.6	0.5	0.0	0.5	0.0	
	FDI share (2000-2001 average)	100.0	25.7 14.9 9.3	2.6	5.4 4.6 4.6	3.8 84.6	2.6 1.7 1 4	1.0	0.9	0.08	0.7 13.6	0.6	0.2	0.1	0.0		
`	GDP share (2001)	100.0	10.0 8.9 4.6		6.3 1.7	$\frac{2.7}{51.8}$	4.6.2 4.1.0	1.40	22.0	1.7 3.3 1.0	3.1 37.4	$1.6 \\ 1.7$	1.0	0.3	1.1	0.1	bina, 2002.
SS)	Population share (2001)	100.0	6.1 5.8 1 3	2.7	0.8 0.8 0.8	31.7	4.7 5.2 2	0.0		2.9 3.0 1	$5.0^{2.1}$	2.4 2.6	2.0 2.0	4.0°	1.5	0.2	earbook of C
percentage	FDI (2001)	46,367	$11,932 \\ 6,915 \\ 4,797$	3,918 3,521	2,516 2,212 2,133	39,207	$1,189 \\ 810 \\ 670$	582 767	457 396 384	352 341 328	337 337 6,322	256 234	107 74 25	300	100 777	1/ - 838	Statistical Y
dollars and	GDP (2001, current price)	1,286	128 115 60	114	61 81 22	34 666	56 48 67	55 - ( 57 - (	276 276	225 43	40 482	21 21	19 13	245	181	139	f Statistics,
millions of	Population (001, million)	1,276	78 74 16	96 7 7 8 7 8	10442 10442	$^{14}_{404}$	60 66 66	98°	44 84 84 84	18/1	63 637	31 33 33	224 264	÷	19 20	226	al Bureau o
(in	Province (2	National Total	Guangdong Jiangsu Shanohai	Fujian Shandong	Liaoning Zhejiang Tianjin	Beijing Top 9 by FDI	Hubei Hunan Hebei	Sichuan Hainan	Henan Jiangxi Guangxi	Shaanxi Heilongjiang	Anhui Middle 12 by FDI	Chongqing Shanxi	Inner Mongolia Gansu	runnan Qinghai	Xinjiang	Tibet Bottom 10 hv FDI	Source: Nation

Table 4.5 FDI and Trade Patterns by Province and Ranked by Provincial FDI Amount in 2001

Borrowing in 2000	ollore and nercentage
China's External	<i>(in thousands of de</i>
Table 4.6	

	percentages)
	and
	lars
-	lob
•	ot
-	nousands
	Ţ
;	II)

Share in Total	29.3	17.7		2.6	20.7	6.6	10.0	0.4	100.0	
Total	4,981,869	3,005,815		437,899	3,519,989	1,124,871	1,704,556	75,324	17,011,323	100.0
Trade Credit									2,161,000	12.7
Portion of Hard Currency Payment elated with Counter- trade					154	10,328			10,482	0.1
Lease		117		591	31,262	976,296		1,378	1,009,643	5.9
Savings by Foreigners		49,764		284			103,375		153,424	0.9
Delayed Payment (Usance?)		176,839			59,498	13,709	20,036	4,821	274,903	1.6
Issuing Bonds	603,039	336,999		249,900	76,998				1,266,935	7.4
Borrowing by exporters, foreign companies and individuals		5,919			2,314,315	25,054		8,718	2,354,006	13.8
Buyer's Credit		1,130,712		4,522	157,978	9,432		2,460	1,305,105	7.7
Foreign Banks	43,935	593,240		182,602	800,517	89,457	1,581,144	57,947	3,348,841	19.7
Interna- tional financial institutions	2,679,027				77,601				2,756,627	16.2
Foreign Govern- ment	1,655,869	712,225			\$ 1,668	596	IS		2,370,358	13.9
Borrower / Type of Loan	State Council, Ministries	Chinese Banks	Chinese Non-Banking	Financial Institutions	Foreign Invested Enterprise	Chinese Enterprises	Foreign Financial Institution	Others	Trade Credit	Share in total

Young Rok Cheong and Geng Xiao 159

Table 4.7 Population, GDP, Savings and Current Account Surplus by Selected Regions and Countries, 2001 (nercentages and hillions of dollars)

	Population	GDP at PPP	GDP (PPP) per	Saving	Investment	Net Lending	Current
	(% of World)	(% of World)	capita as % of the	(% of GDP)	(% of GDP)	(% of GDP)	Account
			World Average				Balance
							(\$ hn)
China	21.0	12.1	58	40.3	37.9	2.4	17.4
India	16.7	4.7	28	20.3	22.9	-2.6	-0.1
Developing Asia	52.2	22.2	43	32.3	30.3	2.0	39.4
Developing countries	78.0	37.6	48	26.8	26.2	0.6	39.6
Countries in transition	6.6	6.2	94	24.0	21.0	3.0	11.8
Newly industrialized Asian Countries	1.3	3.3	254	29.0	23.9	5.1	57.1
Japan	2.1	7.3	348	27.3	25.2	2.1	87.8
European Union	6.2	19.9	321	20.3	20.2	0.1	3.2
United States	4.6	21.4	465	16.5	19.1	-2.6	-393.4

160 Global Capital Flows and the Position of China

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, <u>www.fondad.org</u>

Table 4. 8 China's Balance	ce of Pay	ments, 19	82-2001							
(in millions of c	dollars an	d percentag	ges)							
	1982	1983	1984	1985	1986	1987	1988	1989	1990	1991
Current Account Balance	5,674	4,240	2,030	-11,417	-7,035	300	-3,803	-4,317	11,997	13,270
FDI into China	430	636	1,258	1,659	1,875	2,314	3,194	3,393	3,487	4,366
Net Errors & Omissions	279	117	-932	92	-863	-1,371	-1,011	06	-3,134	-6,748
Reserve Assets Change	-6,291	-4,137	-95	2,353	1,954	-4,931	-2,318	503	-12,118	-14,554
GDP in dollars at average exchange rate				304,423	294,247	321,277	400,000	448,913	389,259	406,443
Accumulated CA since 1982				527	-6,508	-6,208	-10,011	-14,328	-2,331	10,939
Accumulated FDI since 1982				3,983	5,858	8,172	11,366	14,759	18,246	22,612
Accumulated capital flight (E&O) since 1982				444	1,307	2,678	3,689	3,599	6,733	13,481
Accumulated official foreign exchange reserves	6,986	8,901	8,220	2,644	2,072	2,923	3,372	5,550	11,093	21,712
External debt				15,830	21,480	30,200	40,000	41,300	52,550	60,560
Accumulated CA since 1982 as % of GDP				0.2	-2.2	-1.9	-2.5	-3.2	-0.6	2.7
Accumulated FDI since 1982 as % of GDP				1.3	2.0	2.5	2.8	3.3	4.7	5.6
Accumulated capital flight (E&O) since 1982 as % of GDP				0.1	0.4	0.8	0.9	0.8	1.7	3.3
Official reserves as % of GDP				0.9	0.7	0.9	0.8	1.2	2.8	5.3
External debt as % of GDP				5.20	7.30	9.40	10.00	9.20	13.50	14.90

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, <u>www.fondad.org</u>

Young Rok Cheong and Geng Xiao 161

				100 r		1	0	0000		000
	1992	1993	1994	1995	1996	1997	1998	1999	2000	2001
Current Account Balance	6,401	-11,903	7,658	1,618	7,242	29,717	29,324	15,667	20,519	17,405
FDI into China	11,156	27,515	33,787	35,849	40,180	44,236	43,752	38,752	38,399	44,241
Net Errors & Omissions	-8,252	-9,804	-9,775	-17,812	-15,566	-16,952	-16,576	-14,804	-11,893	-4,856
Reserve Assets Change	2,102	-1,767	-30,527	-22,481	-31,643	-35,724	-6,426	-8,505	-10,548	-47,325
GDP in dollars at average exchange rate	481,389	601,223	542,749	701,250	818,873	903,241	960,789	992,353	1,079,481	1,157,211
Accumulated CA since 1982	17,340	5,437	13,095	14,713	21,955	51,672	80,996	96,663	117,182	134,588
Accumulated FDI since 1982	33,768	61,283	95,070	130,919	171,099	215,335	259,087	297,840	336,238	380,479
Accumulated capital flight (E&O) since 1982	21,733	31,537	41,312	59,124	74,690	91,642	108,218	123,023	134,916	139,771
Accumulated official foreign exchange reserves	19,443	21,199	51,620	73,597	105,029	139,890	144,959	154,675	165,574	212,165
External debt	69,320	83,570	92,810	106,590	116,280	130,970	146,040	151,830	145,730	170,110
Accumulated CA since 1982 as % of GDP	3.6	0.9	2.4	2.1	2.7	5.7	8.4	9.7	10.9	11.6
Accumulated FDI since 1982 as % of GDP	7.0	10.2	17.5	18.7	20.9	23.8	27.0	30.0	31.1	32.9
Accumulated capital flight (E&O) since 1982 as % of GDP	4.5	5.2	7.6	8.4	9.1	10.1	11.3	12.4	12.5	12.1
Official reserves as % of GDP	4.0	3.5	9.5	10.5	12.8	15.5	15.1	15.6	15.3	18.3
External debt as % of GDP	14.40	13.90	17.10	15.20	14.20	14.50	15.20	15.30	13.50	14.70
Source: National Bureau of	Statistics,	Statistical ]	Vearbook of C	<i>China</i> , 2002	and Peopl	e's Bank of	China webs	site.		

162 Global Capital Flows and the Position of China

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, www.fondad.org

Table 4.8 (continued)

		An	nount			Share of Gre	ater China Total	
	China	Taiwan	Hong Kong	Greater China	China	Taiwan	Hong Kong	Greater China
1985	9-	-11,697	-5,610	-17,313	0	68	32	100
1986	-1,665	-14,267	-5,861	-21,792	8	65	27	100
1987	-2,796	-17,209	-5,871	-25,876	11	67	23	100
1988	-3,490	-12,585	-4,550	-20,625	17	61	22	100
1989	-6,235	-12,978	-3,431	-22,644	28	57	15	100
1990	-10,431	-11,175	-2,805	-24,411	43	46	11	100
1991	-12,691	-9,841	-1,141	-23,673	54	42	5	100
1992	-18,309	-9,346	-716	-28,371	65	33	3	100
1993	-22,777	-8,934	319	-31,391	73	28	-1	100
1994	-29,505	-9,597	1,745	-37,357	79	26	-5	100
1995	-33,790	-9,682	3,940	-39,532	85	24	-10	100
1996	-39,520	-11,447	4,102	-46,865	84	24	6-	100
1997	-49,695	-12,263	4,829	-57,129	87	21	8-	100
1998	-56,927	-14,960	2,387	-69,501	82	22	-3	100
1999	-68,677	-16,073	2,124	-82,626	83	19	-3	100
2000	-83,833	-16,097	3,133	-96,797	87	17	-3	100
2001	-83,096	-15,253	4,381	-93,968	88	16	-5	100
2002	-103,115	-13,805	3,283	-113,637	91	12	-3	100

 Table 4.9 US Current Account Deficits with Greater China

From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, www.fondad.org

oods in 2002	
ance of G	(00000400
s and Bal	and mono
s, Import	of dolland
US Export	in million
Table 4.10	

percentages)
and
dollars
of
millions
(in

		Amount		S	hare of the Tot	al
	Balance	Exports	Imports	Balance	Exports	Imports
Deal Balance of Documents Pacie	194 252	707 204	1 166 030	100.0	100.0	100.0
IDIAL DATATICE OF I AVIITETILS DASIS	<i>CCC</i> , <b>L</b> 0 <b>L</b> -	000,200	1,100,737	100.0	TUUN	1 UU.U
Vorth America	-86,962	258,360	345,322	18.0	37.9	29.6
Vestern Europe	-89,218	157,080	246,298	18.4	23.0	21.1
Eastern Europe/FSR	-8,283	6,599	14,883	1.7	1.0	1.3
Pacific Rim Countries	-215,005	178,561	393,567	44.4	26.2	33.7
Australia	6,606	13,084	6,478	-1.4	1.9	0.6
China	-103,115	22,053	125,168	21.3	3.2	10.7
Japan	-70,055	51,440	121,494	14.5	7.5	10.4
Newly Industrialized Countries(NICS)	-22,073	69,823	91,896	4.6	10.2	7.9
Hong Kong	3,283	12,612	9,328	-0.7	1.8	0.8
Korea	-12,979	22,596	35,575	2.7	3.3	3.0
Singapore	1,429	16,221	14,793	-0.3	2.4	1.3
Taiwan	-13,805	18,394	32,199	2.9	2.7	2.8
Other Pacific Rim(3)	-26,369	22,162	48,531	5.4	3.2	4.2
South/Central America	-17,902	51,643	69,544	3.7	7.6	6.0
DPEC	-34,482	18,852	53,334	7.1	2.8	4.6
Other Countries	-36,397	28,956	65,353	7.5	4.2	5.6

164 Global Capital Flows and the Position of China

Source: US Census Bureau website.

		Outside Banking	Outside Banking	Net Outside
	D : 1	Claims on HK	Liabilities to HK	Banking
	Period	(HK Gross	(HK Gross	Liabilities to HK
		Borrowing)	Lending)	(Net HK Lending)
	1997-06	620 554	640 490	10.036
All Outside	2002-04	249 541	389 987	140 446
Hong Kong	Change	-380.013	-250 504	129 509
110118 110118	Change in %	-60.4	-39.1	1184.2
	1997-06	309.067	361 423	52 357
	2002-04	57 942	81 484	23 542
Japan	Change	-251.125	-279.939	-28.814
	Change in %	-81.3	-77.5	-55.0
Singapore	1997-06	48 863	37 597	-11 266
	2002-04	24 214	39,700	15 486
	Change	-24.649	2,103	26.753
	Change in %	-50.4	5.6	-237.5
	1997-06	50.679	28.231	-22.448
	2002-04	18,753	60.207	41.454
UK	Change	-31.926	31,976	63,902
	Change in %	-63.0	113.3	-284.7
US	1997-06	24,974	23.606	-1.368
	2002-04	12,491	31,806	19.315
	Change	-12,482	8.201	20.683
	Change in %	-50.0	34.7	-1511.7
	1997-06	40,087	50,105	10,018
Mainland	2002-04	39,312	16,651	-22,661
China	Change	-776	-33,454	-32,679
	Change in %	-1.9	-66.8	-326.2

Table 4.11	Cross-Border Banking Capital Flows in Hong Kong, 1997-20	02
	(in millions of dollars and percentages)	

Source: Hong Kong Monetary Authority website.



#### Figure 5.1 FDI Inflows, 1979-2001 (in billions of dollars)

Figure 5.2 US Current Account Deficits with Greater China (in billions of dollars)



From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, www.fondad.org



Figure 5.3 Banking Capital Flows between Hong Kong and Mainland China



From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, <u>www.fondad.org</u>



From: China's Role in Asia and the World Economy - Fostering Stability and Growth, FONDAD, December 2003, <u>www.fondad.org</u>

#### 170 Global Capital Flows and the Position of China

#### Figure 6.3 The Lower the Local Turnover, the Higher the Share of Trading by US Residents, The Case of Japan (in percentages and billions of dollars)



#### Figure 6.4 The Lower the Local Turnover, the Higher the Share of Trading by US Residents, The Case of Hong Kong (in percentages and billions of dollars)



#### 172 Global Capital Flows and the Position of China

#### Figure 6.5 The Lower the Local Turnover, the Higher the Share of Trading by US Residents, The Case of Mainland China (in percentages and billions of dollars)



#### Figure 6.6 The Lower the Local Turnover, the Higher the Share of Trading by US Residents, The Case of Singapore (in percentages and billions of dollars)



#### 174 Global Capital Flows and the Position of China

#### Figure 6.7 The Lower the Local Turnover, the Higher the Share of Trading by US Residents, The Case of Taiwan (in percentages and billions of dollars)



#### Figure 6.8 The Lower the Local Turnover, the Higher the Share of Trading by US Residents, The Case of Korea (in percentages and billions of dollars)



## 9

### Comment on Young Rok Cheong and Geng Xiao

Li-Gang Liu\*

This overview paper is very thought-provoking. It has raised many important issues facing the Chinese and world economies. The authors have painstakingly compiled a wealth of data, as indicated in 38 tables and charts, to provide evidence for their argument. The paper raises many interesting issues – each of which would be sufficient for a paper itself; but it is also styled so well that we can understand these complex issues with ease.

Let me give a quick review of key points of the paper. Essentially, the authors raise nine questions.

The first one is how to reconcile the fact that China is exporting capital and, at the same time, importing a large amount of capital. The answer they provide is that China used its trade surplus to finance its main customers, mainly the US customers.

The second question is whether China is attracting too much FDI. The answer appears to be 'yes'.

The third question is what happens to China's investment in US bonds. The authors' figure indicates that China has become the second largest bondholder of the US treasury. Indeed, by 2001, China purchased the same amount of bonds as Japan.

<sup>\*</sup> The discussant was formerly a senior research fellow at the Asian Development Bank Institute in Tokyo. He now is an assistant professor of Public Policy and Finance at the School of Public Policy of the George Mason University in Arlington, Virginia, USA.

The fourth question is: What is the impact of foreign portfolio investment on the stability of the Chinese stock markets? At this moment, the impact is negligible given that China's capital account is closed, but we may expect more impact in the near future as the Qualified Foreign Institutional Investors (QFII) plan to allow foreign institutional investors to set up joint-venture mutual funds in the Chinese stock market.

Fifth, they ask if China is generating global deflation. The answer is 'yes' in the labour-intensive manufacturing products but 'no' in overall manufactured products since China's share in world trade is still rather small, around 4 percent in 2000.

Sixth, they are interested in determining the seriousness of China's impact on its competitors. As Geng Xiao mentioned in his presentation, there are both benefits and costs of China's emergence in the world economy. Perhaps the rational strategy is for other countries to respond proactively and deal with the challenge of the emergence of the Chinese economy.

Seventh, they raise the issue of whether China is saving too much. Their answer is affirmative. Demographics and growth can explain this phenomenon. However, Chinese savings are not well utilised. The existing financial system is the root problem.

Eight, given large capital inflows and domestic savings, they ask a hotly debated question: Should China revalue its currency? Their qualified answer is, 'No, but perhaps some domestic price adjustment should be used.'

The last question they entertain is whether China has become a growth engine for the world. From the paper the answer is not that clear. Although China has become a major trading nation in the world and its GDP size in PPP is the third largest in the world, it is not obvious that China is an engine of global growth at this moment.

Having recapped the key points of the paper, I would like to make the following observations and comments.

The first one concerns the sustainability of the current trade-FDI pattern in China. There is a big difference between 'made in China' and 'made by China'. At present, more than 50 percent of Chinese exports is conducted in the form of processed trade: China imports intermediate components, mainly from Japan, South Korea, and Taiwan, and then assembles them for exports. The assembled products are then disproportionately exported to the US market.

This pattern of triangular trade has allowed China's powerful exporting neighbours, South Korea, Taiwan, and to some extent, Japan, to divert their previously US-bound exports to China, thereby reducing their trade surplus with the US. If this triangular feature of the China-US trade were to be taken into consideration, the adjusted real trade balance between China and the US would be far smaller than the current number since China's value-added in the processed trade has been rather minimal, mostly in the form of low wages of the assembly workers. Indeed, this feature is evidenced by the fact that although China has a large trade surplus with the US, it runs a similarly large trade deficit with South Korea, Taiwan and Japan. Therefore, China's global trade surplus is small, accounting for only 2 percent of its GDP per year. As the US-China trade deficit balloons, it is not surprising that China has become a new target of US trade policies, similar to Japan and other East Asian economies in the 1980s. Thus, it is very doubtful whether the current pattern is sustainable and whether China can continue to export its trade surplus to the US. At this moment, China's trade-to-GDP ratio is close to 50 percent and the US current account deficit is historically high, possibly reaching 7 percent of its GDP by the end of this year. One would have to wonder whether this size of US current account deficits could be maintained. There will have to be some kind of adjustment on both sides.

Second, I am puzzled by the fact that the wage rates in the export sector have been rather stagnant over the last ten years despite an impressive improvement in the sophistication of Chinese exports. The authors mention that FDI-funded firms are the main drivers of China's export growth contributing to 50 percent of China's exports. Indeed, China has become the largest FDI recipient in the world for the first time this year. Then it is puzzling that the wage rates in the exporting sector have not gone up much over the last ten years. Empirical evidence indicates that, normally, a country with a large amount of FDI like China would experience a rapid growth of wages in the export sector. The authors mention that in Guangdong province, over the last ten years, the average wage for manufacturing workers is only 100 dollars per month. I suspect that this rather stagnant wage pattern has a direct connection with the ongoing restructuring in the state-owned enterprise sector and the unlimited supply of labour from China's vast rural area. Because of these two factors, the wages in the export sector have not led to an overall

178

domestic wage growth, as is usually observed in a country with large inflows of FDI. However, without wage growth, it will be difficult for China to raise domestic demand. This is perhaps why China's growth is so dependent on export growth, as indicated by its tradeto-GDP ratio.

If you look at the nominal GDP among China, the US and Japan over the last 30 years, China's GDP in nominal US dollars as a share of United States' GDP has not changed much; it stayed more or less below 10 percent of United States' nominal GDP. On the other hand, if you look at Japan, its GDP as a share of United States' GDP has converged to the size of the US GDP until the bubble collapsed in the 1990s.

It is quite puzzling that although China has experienced quite high growth, it has not grown much in terms of nominal value of US dollars. This seems to indicate that the wealth from FDIgenerated income is located overseas rather than within China. I wonder whether this is because the net value added in China's exports has been minimal, mainly in the form of cheap labour income.

The third issue I would like to comment on is whether there is an FDI diversion from ASEAN to China. This is a controversial issue, especially among Asian countries. However, I want to put this in perspective by asking whether the current trend of high FDI in China is a cyclical issue or a permanent phenomenon. If you look at the overall FDI flows to less-developed countries, excluding China, over the last several years, they have increased. Why have ASEAN countries received less FDI from the rest of the world? I think domestic factors play the major role. ASEAN countries were hit by the financial crisis of 1997-98 and the recovery from the crisis was further exacerbated by the September 11th terrorist attacks in the US. In addition, Islamic countries such as Indonesia and Malaysia may have suffered from a higher investment premium. Western investors may view these countries as quite risky, so they prefer not to move their assets to these countries for the time being.

If we look at the adjusted per capita flows and stock figures of FDI, as the two authors mention, 25 percent of FDI from Hong Kong to China is round-tripped FDI. That is, Chinese money or investment is first taken to Hong Kong or other offshore facilities and is then taken back to China in order to enjoy tax incentives, favourable land use concessions, better property protections, and

other types of subsidies. If these tax incentives are going to be phased out with China's WTO commitment, I feel that such capital flight may not return. After the dust has settled in the ASEAN economies, perhaps such capital from China will go to Malaysia or Indonesia, rather than go back to China.

The fourth issue I would like to discuss is an interesting and rather controversial one, that is, the prediction of the Lewis model the authors used: If the Lewis model is applicable, China is going to contribute to world deflation, or wage deflation at least. According to the authors, currently 224 million of Chinese would be participating in the world market, while two-thirds of the population are still waiting in line. If we think those people are waiting in line to get into the labour-intensive industry only, perhaps that kind of prediction could materialise. However, if we qualify this assumption by assuming that China will continue to grow with industrialisation and urbanisation, the non-tradable service sector will increase to a normal size as we have seen in most of the industrialised countries, and, as a result, the immiserising impact of Chinese export growth on the rest of the world will not take place.

Thus, the participation ratio of Chinese workers in the world economy will be much lower than the authors' prediction, and it is likely that a majority of people in China will engage in the nontradable service sector rather than compete internationally. Also, as I mentioned, China cannot maintain a 50 percent trade-to-GDP ratio forever. This ratio is obviously unsustainable for a continent-sized country like China.

However, the issue is, as China's nominal GDP ratio to Japan and the US shows, that China's terms of trade have been flat over the years, despite the fact that it has been climbing the value-added product ladder. Perhaps the average wage rigidity, due to a lack of active bargaining power of the workers in the foreign-owned firms, might be the reason why wages cannot go up. And perhaps because of this problem, the income inequality in China in the 1990s has become worse. The problem is that local governments have great incentives to attract FDI, so in a sense, they tend to collude with foreign investors and the interests of workers may have been ignored here.

What is the policy implication then? Wing Thye Woo has mentioned that one should try to increase the income of the rural sector. I think we should go a bit further and also look at the wage issue of FDI-funded firms. The question is whether the bargaining power of workers can be increased so that their wages can be increased through the bargaining process. In fact, this will be one of the solutions if China wants to sustain its growth by creating sustainable domestic demand in the future. Almost 90 years ago, when Henry Ford offered his workers a wage of 5 dollars a day, a much higher wage than usual, he had an economic insight not apparent to most people at the time. If a majority of his workers could not afford to buy his Model-T automobiles, who would demand them after productivity and efficiency were improved so immensely that the mass production of cars was no longer a fantasy? His insight proved to be right.

Today, the world is facing a similar challenge: integrating large, populous and poor developing countries into the world market. If this succeeds, it will boost their manufacturing capacities because of their large pool of semi-skilled industrial workers and, at the same time, it will directly impact the welfare of industrial workers of the developed world. To address this challenge, the wage rates and income of those developing countries will have to go up so that their own large internal markets can demand some of the goods produced. Besides waiting for the labour market alone to adjust the wage rates in developing countries, this process can also be facilitated if governments in both developed and developing countries collectively pay attention to and enforce their labour laws so as to allow wage bargaining between workers and capitalists to take place. On the other hand, the beneficiaries from the globalisation process, mostly the transnational corporations, will have to share some of their profits with the industrial workers in the developing countries. Similar to the insight of Henry Ford, it may be a difficult policy for some to accept at the moment. But for the sake of global prosperity, such measures will prove to be beneficial for everyone. Presumed as a champion for the welfare of workers and peasants, the Chinese government can play a decisive role in initiating a global movement that will benefit its own millions of industrial workers and peasants at home.

# 10

## Floor Discussion of "China's Role in the Region and in the Global Financial System"

#### The Park and Wang Paper

Masaru Yoshitomi, former dean of the Asian Development Bank Institute (ADBI), wondered whether China's economic boom would eventually end in a financial crisis and what could be done to prevent such a crisis from occurring. "The main reason for the Chiang Mai Initiative is that we want to prevent a crisis similar to the capital account crises that struck Asia in 1997, and to better manage such a crisis when it would occur. First of all, we have to understand the real nature of the capital account crises of 1997. At the ADBI we have just released a report<sup>1</sup> that develops a set of policy recommendations for China on how to sequence its financial liberalisation, given that the sequencing order of financial liberalisation was clearly the missing link when the World Bank published its Asian miracle report in 1993 and only four years later the Asian crisis erupted.

The ADBI found that there was a huge gap between the new risks arising from financial liberalisation and deregulation (changing the incentives of borrowers and lenders), on the one hand, and the pre-existing, old kind of institutions including prudential regulation and supervision, on the other hand. The old regulatory frameworks

<sup>&</sup>lt;sup>1</sup> ADBI, "Policy Proposals for Sequencing the PRC's Domestic and External Financial Liberalization", October 2002.

and institutions were clearly not prepared for managing the new risks under liberalisation because they had no idea about how to do capital account liberalisation. That is quite natural because under the old, regulated financial markets we did not need such concepts at all.

The capital account crisis did, in fact, happen in the 1990s in Asia, and it might happen in the future in China as well. I have come to the conclusion that big financial crises usually occur after a real economic boom with some technological improvements. In this decade, the Chinese economy will continue to boom at least until we have the Olympics in 2008, and probably until 2010 when we have the World Expo, financed by the banks and so on. The WTO accession will liberalise financial services transactions in China by inviting all kinds of financial institutions. So at the end of this decade, there is likely to be a risky combination of a continued economic boom together with a real estate bubble financed by the banks, not by the securities market like in Japan in the 1980s.

The *de facto* capital account liberalisation in China will probably end up in a capital account crisis because such crises tend to take place, not every year but every ten or fifteen years. In order to prevent a crisis from occurring in China, besides implementing the kind of policy recommendation on the sequencing order of financial liberalisation that we at the ADBI have suggested, we may also need the arrangements beyond the Chiang Mai Initiative that Yung Chul Park and Yunjong Wang have suggested in their paper. In my later panel presentation I will talk more about the kind of recommendations we have made in our ADBI report." (See Chapter 17 of this volume.)

Xie Ping, of the central bank of China, thought that it was not yet clear to the Chinese government how regional financial cooperation beyond the Chiang Mai agreement should be developed. "It is not clear because China receives too many different suggestions. One day the ADB makes a suggestion, one day the World Bank, one day the IMF, and another day the BIS. It has been suggested, for example, that we create an Asian bond market, and that such a market would be very useful for the Asian countries including China. In the context of the Chiang Mai Initiative, last year we agreed on currency swaps with Japan, Thailand, Korea and some other countries, amounting to about 10 or 20 billion dollars. Some people say that this amount is very small and that, if China would face a financial crisis, it would be absolutely nothing. So maybe this currency swap agreement is just symbolic. It may just mean that China participates actively in efforts at East Asian financial cooperation. Also, we have been discussing this Asian Monetary Fund proposal for a very long time. Some people are upset with it and are saying, 'You cannot do this!', while others are saying, 'This is a good idea'. There are so many people with different opinions about the Asian Monetary Fund.

Last week Professor Mundell from Columbia University, was in Beijing for a week. I met with him and he suggested to me that in the next 10 years, Asia should be united with one currency, and that this currency should be based on the Chinese yuan and the Japanese yen. He also thought that the Chinese yuan should be fully convertible before the 2008 Olympic Games. He gave us, the central bank, a lot of suggestions."

Wing Thye Woo, of the University of California, said that in every regional or global arrangement there is always an issue of economic power involved. "Look at Japan's experience in the IMF. For many years, the Japanese tried to enlarge the capital base of the IMF to increase Japan's voting power, but the Americans and the others resisted it. So any kind of regional financial institution set up now in Asia would mean locking in the present distribution of economic power. The president of such an institution would possibly be a Japanese, the executive vice-president a Chinese, the deputy vice-president a Korean. But let us say that we wait 20 years from now, perhaps the president and the next 5 executive vicepresidents would be all Chinese. So what's the incentive for China to lock-in the present status-quo distribution of power?

Another question is why China should be in favour of a regional bloc when its trade relationship is global in nature. It should be in favour of an Asian bloc only if this would be a building block toward multilateralism. Then the Chinese would be more enthusiastic about it. If you look at China's aspirations for the future, I think it would see its interests better served by a multilateral world rather than a regional bloc; and if it has to be a regional bloc, it should be seen as a bargaining chip to push the multilateral agenda. That's why I think the Chiang Mai Initiative has no life beyond monetary cooperation."

Geng Xiao, of the University of Hong Kong, agreed with Wing Thye Woo on the economic power issue in the Chiang Mai Initiative. "The difficulty of the Chiang Mai Initiative sort of effort

is that, at present, there are no credible leaders. Japan has a problem with its financial system, and nobody knows how to fix it. China is basically free-riding on the US system, pegged with the US dollar and trading with the US. So it is indeed very important to go back to fundamentals, as Yoshitomi and the ADBI people have done, and look at the so-called Asian miracle. If you look at Japan. Korea. Singapore, Hong Kong and all of these countries, they are all special cases of Asia in the sense that they are all small countries with a limited amount of labour supply and when the growth starts, the labour supply hits the limit and then wages go up. But in Asia, which has 50 percent of the world's population, the typical Asian situation is the case of unlimited supply of labour, fitting into the traditional Lewis model, the dual sector model. When China started growing, we had a huge population at subsistence level wages and an unlimited supply of labour, and that had an immediate effect on the ten years of growth. If you look at the Asian countries, we have all these successful economies of Japan, Korea, Taiwan, and now China, and you always see these huge current account surpluses, which mean that these countries are exporting capital, are having huge savings. But at the same time we have a huge surplus of labour in Asia. This means that we have a surplus labour economy exporting capital to the capital-rich US economy instead of investing it in the unlimited supply of labour economies in Asia. If capital and labour combine in Asia, these two things will continue to drive Asian growth. But the problem now is that the capital cannot go to Asia, it can only go to the US, and from the US it goes back to Asia through FDI. So we are actually free-riding on the US system, largely because the US financial system is the most competitive system in the world.

So in this sense, we have to go back to these fundamentals, to assess whether the free-riding on the US system is the best solution for Asia. Hong Kong is the typical case, it is just free-riding, and there is no monetary policy. If Japan would resolve its banking problem, the Japanese interest rate would rise, which would help to better allocate resources because at a 0 percent interest rate, this capital just does not know where to go, it just goes to the US.

We are all without choice just investing in the US, but at the same time we have so many resources in terms of labour, in terms of oil in Indonesia, all kind of resources. We have technology in Japan and in Taiwan, and we have financial markets in Hong Kong, but we are not using it. Last year, even Hong Kong had capital flight to the US. In Asia there is no leader, we are still relying on the US. But we have to ask ourselves whether that is the best solution."

Yung Chul Park was not sure that China would face a financial crisis in the next 10 to 15 years. "Whether China is going to be the next epicentre of a financial crisis within the next ten years or so depends upon a couple of assumptions. The first is, to what extent China is going to borrow from abroad, from the global market. The second is, to what extent China is going to borrow from other Asian countries. In the next 10 to 15 years, I don't think China is going to borrow so much from outside. Since the savings rate is still very high, China will continue to run a current account surplus for some time. But the foreign debt of China will not go up to the level of Korea or Thailand or Indonesia, so the foreign debt-to-GDP ratio will remain very low.

I am more concerned about domestic financial turbulence in China because of this non-performing loan problem and the bankruptcy of the state-owned banks. And this domestic instability may spill over to other East Asian countries and that might be something to worry about even though I think the spillover would be relatively small. So I am not as much worried about the possibility of China getting into a serious financial crisis as Yoshitomi."

Li-Gang Liu, of the ADBI, was puzzled about the leadership issue in Asian financial cooperation that Park and Wang raised at the end of their paper. "I am not sure that the analogy of Germany and France, as leaders in European integration, necessary applies to China and Japan as leaders of Asian integration. In terms of economic size, China is just one-fifth of the Japanese economy. Also, if the rest of the Asian countries are further advanced in terms of financial services liberalisation, capital account liberalisation and domestic financial liberalisation, why can't they start some kind of a financial cooperation framework? China can always join at a later stage. The IMF system started without Russia, and the WTO did not have China as a member for many years, so why should the other countries, if they are willing to go ahead, wait for China?"

Zdeněk Drábek, of the WTO, thought however that Chinese cooperation was of key importance. "Since the stability of the exchange rate in the region is important, the Chinese role in that context is important. It would be very useful for Korea, for Thailand, for Indonesia to know whether the Chinese are going to devalue their currency or not. So in that sense monetary cooperation with the Chinese is not premature but highly important."

In his reply to the comments, Yunjong Wang stressed that the basic aim of the Chiang Mai Initiative (CMI) and the Asian Monetary Fund (AMF) is to prevent financial crisis and manage the crisis better. "In terms of crisis prevention, we need two things: one is the swap facility and the other is the monitoring and surveillance system. Regarding the facility, the current CMI just links to the IMF, so if we are really satisfied with the current Chiang Mai swap scheme, we should probably be happy with the way in which the IMF deals with crisis prevention and management as well. Then we don't need to go any further. But if we are dissatisfied with the role of the IMF and the current scheme of the CMI, then we should think about what kind of elements could be introduced to better prevent the crisis. Regarding the monitoring and surveillance system, the main task is to identify the emerging problems in East Asia. The policy dialogue has already started, but that's not enough, we need a secretariat to detect and initiate a discussion about the emerging issues.

Probably ASEAN+3 will just have to forget about exchange rate coordination, because it is too early. Currency union and currency unification is a delicate political issue. In terms of exchange stability, European countries have a different past, they started with the fixed exchange system, and they didn't need to care too much about exchange rate stability. The Bretton Woods system guaranteed exchange rate stability through various ways, and for two decades the European countries had quite stable exchange rates and also capital controls.

In East Asia there are now different degrees of capital market opening. Geng Xiao talked about China's free riding, but I think that China's monetary policy is independent because China has a restrictive capital market opening. Hong Kong has a currency board system and its capital market is open, so it is very sensitive to US monetary policy. But China can maintain an independent monetary policy.

Talking about currency unification, there is indeed this fact of status quo in economic power. It is a very important factor. If we pursue monetary unification, political will is important. Political scientists are trying to explain why the East Asian countries have no particular political will. European leaders have a kind of new functionalistic vision underpinning the European system. In Europe there is a lot of interaction between the countries and the spillover effects are large, so there is a natural need for policy coordination. But in East Asia we still have a relatively low degree of intraregional trade. It is increasing but it is still low. We also have minimal financial integration among the countries, but that is a less important factor. The most important factor is that countries have different policy objectives. For China, economic development is a more important issue than regional integration, and Japan may have yet another policy objective. Different political instruments are required for these different policy objectives. My conclusion is that it is probably still too early to say much about a high level of economic integration in East Asia."

#### The Cheong and Xiao Paper

Robert McCauley, of the BIS, wondered whether the Lewis model, in the modern monetary world, would be the right way of looking at wages and employment in China. "The current world is quite different from the employment model when potential growth was reasonably well defined. For example, if the renmimbi were to revalue, would that mean that the nominal wage would simply go down more or less in proportion to the devaluation?"

Geng Xiao explained that in the Lewis model, the real wage is determined by subsistence labour and the unemployment rate. "There is a lot of unskilled labour in China, and labourers are competing with each other to find a job. On the one hand, the wage level has to be higher than their incomes in the countryside and, on the other hand, they have to face the unemployment in the cities. So these two conditions more or less determine the wage rate. Chinese people have lived so many years of endurance and frustration in real income that they can deal with low wages. An exchange rate revaluation would probably have an effect on real wages, but not too much. On the longer time horizon of 5, 10, 20 years, wages are hard to change, largely because you have so many people coming out of the countryside who are looking for a job. Wages will go up largely when things become cheaper and the living standard increases. Food has become cheaper for everybody in China, including the people in the countryside, and that has increased their living standards."

Masaru Yoshitomi wondered how one could reconcile the fact that income in China has increased and, at the same time, a large part of the population still lives at a subsistence level. "How does the Lewis model apply to China? On the one hand, we are hearing that the income of people in China has increased over the past 20 years and that it will continue to increase in the next 20 years. But, on the other hand, Geng Xiao insists that the Lewis model applies because the wage level will continue to be based on wages at the subsistence level. How do you reconcile these two sort of conflicting facts?"

Young Rok Cheong thought the explanation was quite simple. "These facts go together because there is a large inequality in income in different areas. Look at the per capita GDP in the cities, and at the difference between the top rich 9 cities and the rest. The gap is widening."

Yoshitomi insisted: "But the question is: Have the wages of the labourers who migrated to the coastal areas remained constant over the past 20 or 30 years? And are the wages in the coastal areas still determined by the subsistence level in the very remote areas in China? That is what I'm asking."

Cheong explained: "You will be surprised to know that wages for the workers in Guangdong in assembling lines in foreign investment enterprises are actually lower than probably in the whole nation because the competition in the labour market is so perfect that labourers are coming from all over China. If you want to find a job you go to the coastal cities. The wages there are not higher than in the interior cities."

Yoshitomi continued: "What kind of immigrants are they? Are they coming into the cities from relatively near or from remote areas?"

Cheong: "Both, they are coming from the Western part of China, and they are coming from within the province. Many of the migrating workers are not seeking the job by themselves but are being sent by provincial governments. They organise the provision of cheap labour, as cheapest as possible, and export the labourers to the coastal areas. I visited many companies and they have their own dormitories and maintain a very low level of wages."

Geng Xiao added that for skilled labour, wages are increasing rapidly in China. "The wage income of the professors and the technicians is going up very fast, every year. The gap between skilled and unskilled labour is increasing rapidly. Wages for unskilled labour, on the other hand, remain low because you have plenty of supply. For example, when you advertise for a waitress, there are hundreds of people applying. How can you then increase the wage? Even in Hong Kong the wage for unskilled labour has not changed much, it is still the same as 10 years ago."

Charles Adams thought that at the time Lewis developed his model, he did not make a distinction between product wages and consumption wages and that the subsistence argument is about the consumption wages. Second, he found it very odd to be talking about China's exporting deflation. "At best it is an issue of relative prices. But in the case of the US, manufacture prices have been rising less rapidly than service prices for a long time, and well before China was potentially on the scene. So I suspect it has something to do with differential productivity. And just to make a provocative point: while it seems that one can make the case that the RMB does not need to appreciate vis-à-vis the US dollar, one can also make the case that the RMB should appreciate against the rest of Asia. What would happen if the US dollar weakened against the other Asian currencies and the RMB did not change?"

Geng Xiao said that his quick answer would be that most Asian economies are pegged to the US dollar, "so when the US dollar changes everyone just follows".

Adams observed that there might be a *de facto* pegging, but on paper, most countries were floating. "So if we see adjustments in other Asian currencies, then there will be an issue about the crossrate between those currencies and the RMB. So all I'm asking you is, if you think on that margin, what would be your answer?"

Xiao: "According to economic theory it is very simple: you just don't mess with the exchange rate. The issue of prices and price changes is confusing in China because the country is not an equilibrium economy. The exchange rate is about inflation; it is about the price. In China, there are many sectors and many regions, all fragmented, and they are all trying to find equivalent prices. So the exchange rate is used as an anchor."

Adams: "I may be old-fashioned, but nominal exchange rate changes have quite a profound effect on real exchange rates, for a sustained period, in many countries."

Xiao: "That is true, but exchange rate fluctuations in China would also have a huge wealth impact in the sense that China's holding of foreign assets are huge. I did some rough calculations:
once you change the exchange rate by, say, 10 to 20 percent, there is going to be a redistribution of wealth in China of the range of 7 to 8 percent of GDP. That is why I doubt whether China should have floating exchange rates. The monetary system in China relies so heavily on the US dollar, 45 percent of China's central bank assets are in US dollars, and 24 percent of China's GDP is in US dollars. So if you mess with the exchange rate, you will stimulate huge speculations of people who will be trying to make money out of the exchange rate changes. That's why the exchange rate in China should remain fixed. By remaining fixed, the renmimbi remains credible – at least, as long as China continues to have a surplus. And a fixed rate substantially reduces the risks for traders, producers, foreign investors and Chinese people. So, in my view, the current monetary system is the best for China. I don't believe in the traditional IMF recipe because that is only right for an economy based on an equilibrium model. It is better to give China an exchange rate anchor, so that it can fully employ its labour force. Once that is realised, we can talk about floating and all the other traditional arguments coming into effect. But first China has to address the basic question of development."

Xiao stressed that China's trade-to-GDP ratio of 45 percent was rather meaningless because of the minimal value-added in China's exports, and that China would remain a very attractive country for foreign investors. "As a percentage of GDP, the value-added is very small and China's GDP, in PPP terms, is highly undervalued. It is the opposite in Japan. Japan's exports are only 10 percent of GDP because its GDP, in PPP terms, is overvalued. I don't think that the potential of FDI into China is going to decline, largely because once every major multinational corporation has production facilities in China, you generate an economy of scale and scope that is going to have its own momentum. Look at what happened earlier in history in Manchester, New York, Chicago, Tokyo, Osaka – they have all become major manufacture centres of the world. The same is now happening in Shanghai and in Guangdong."

Yunjong Wang shared Xiao's view that China would continue to attract foreign investors, because of its own large population and that of neighbouring India. "I cannot see a turning point in foreign investment in the near future. India, with its large English speaking population, is trying to establish special economic zones along its coast. Just like China, India also has an unlimited supply of unskilled labour. The two countries together may have 1 billion unskilled labourers. That is going to attract a lot of capital. But we should be careful. There are still many foreign investors who do not want to invest in the western part of China. To attract foreign capital, you need a good infrastructure and a good working discipline. Foreign factories are mostly concentrated on the coastal line and this might create growing inequality, undermining social cohesion and stability. However, both India and China have an enormous potential for foreign investments."

Geng Xiao emphasised that the competitive pressure of China in the world is just there, not only because of the Lewis model and the unlimited supply of labour, but also because of the mobility of institutions and of capital. "The successes of Shanghai and Hong Kong are tremendously important. You cannot believe how many people are focused on China, and how much advice is given to China. Look at the whole world, look at the major US and European corporations, or at the major international, US and European institutions, they all have China experts who study China, advise China, and think about China. All the high techs are in Hong Kong. So this is real. If China continues its success, the implications for the whole world are tremendous. It is so simple, I mean, 1000 or 500 years ago we spent most of our time on getting food and shelter. but today one hour is for food and the other time is for entertainment. In a way, this relative price adjustment is good news, it means that now, in principle, it should be easy for everyone in the world to have a decent basic life. We have the technology, the knowledge and the institutions to realise that."

Zdeněk Drábek wondered about the importance of FDI for China's growth. "How important is the foreign sector? What is really driving the growth? I would think that most of the aggregate demand is domestic and that most of the investment must be domestic investment. So I don't think that foreign investment is the engine of economic growth in China. What is much more important is what the government is doing. I think that the fiscal policy and the investments by state enterprises are driving this big investment drive in China."

Young Rok Cheong agreed. "The magic is not just FDI or the unlimited supply of labour. The magic is to put everything together. The Chinese government is a very stable government and very open-minded to foreigners. And indeed, the infrastructure investments are huge and unprecedented in China."

Cheong ended the discussion with three policy suggestions. "The first is that the Chinese government can and should allow the market to work. The market is working at the global level and it should also work at the national level. Second, once China fully employs all its surplus labour, it can apply the traditional recipes of economic equilibrium theory. China's labour force provides competitive pressure to other countries, but it also harbours domestic risks, social risks. Third, the development of the domestic financial system in China is of key importance for sustained growth. Economic growth is facilitated by the protection of property rights, and the protection of property rights in China is coming through in the wrong way, through the foreign investment enterprises. Only when the foreigners came in, the government started to secure property rights and apply other standards of international practice in, for instance, the hiring of workers. Property rights protection is going to be the key for China to have success in the next stage. The financial sector is a derivative of the real sector and anything that goes wrong will be reflected in the financial sector. So by tracing the road for the financial sector in China, we can trace the road for maintaining growth and development."